

# Tectonics

## **RESEARCH ARTICLE**

10.1029/2018TC005036

#### **Key Points:**

- Coeval India-Asia collision and Pacific back-arc extension explained by asthenospheric flow
- Mantle flow correlates with long-term kinematics from India to Pacific border
- Indian mantle reaches Pacific border and Asian lithosphere spreads above ponding slabs

Supporting Information: • Supporting Information S1

Supporting information 5

#### **Correspondence to:**

L. Jolivet, laurent.jolivet@sorbonne-universite.fr

#### Citation:

Jolivet, L., Faccenna, C., Becker, T., Tesauro, M., Sternai, P., & Bouilhol, P. (2018). Mantle Flow and Deforming Continents: From India-Asia Convergence to Pacific Subduction. *Tectonics*, *37*, 2887–2914. https://doi. org/10.1029/2018TC005036

Received 26 FEB 2018 Accepted 1 AUG 2018 Accepted article online 7 AUG 2018 Published online 6 SEP 2018

#### ©2018. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

## Mantle Flow and Deforming Continents: From India-Asia Convergence to Pacific Subduction

Laurent Jolivet<sup>1</sup> [D, Claudio Faccenna<sup>2</sup> [D, Thorsten Becker<sup>3,4,5</sup> [D, Magdala Tesauro<sup>6</sup>, Pietro Sternai<sup>7</sup> [D, and Pierre Bouilhol<sup>8</sup> [D]

<sup>1</sup>Sorbonne Université, CNRS-INSU, Institut des Sciences de la Terre Paris, ISTeP UMR 7193, Paris, France, <sup>2</sup>Dipartimento Scienze, Università Roma Tre, Rome, Italy, <sup>3</sup>UTIG, University of Texas at Austin, Austin, TX, USA, <sup>4</sup>DGS, University of Texas at Austin, Austin, TX, USA, <sup>5</sup>JSG, University of Texas at Austin, Austin, TX, USA, <sup>6</sup>Dipartimento di Matematica e Geoscienze, Universita degli studi di Trieste, Trieste, Italy, <sup>7</sup>Department of Earth Sciences, University of Geneva, Geneva, Switzerland, <sup>8</sup>C.R.P.G., Université de Lorraine, Vandoeuvre-les-Nancy, France

Abstract The formation of mountain belts or rift zones is commonly attributed to interactions between plates along their boundaries, but the widely distributed deformation of Asia from Himalaya to the Japan Sea and other back-arc basins is difficult to reconcile with this notion. Through comparison of the tectonic and kinematic records of the last 50 Ma with seismic tomography and anisotropy models, we show that the closure of the former Tethys Ocean and the extensional deformation of East Asia can be best explained if the asthenospheric mantle transporting India northward, forming the Himalaya and the Tibetan Plateau, reaches East Asia where it overrides the westward flowing Pacific mantle and contributes to subduction dynamics, distributing extensional deformation over a 3,000-km wide region. This deep asthenospheric flow partly controls the compressional stresses transmitted through the continent-continent collision, driving crustal thickening below the Himalayas and Tibet and the propagation of strike-slip faults across Asian lithosphere further north and east, as well as with the lithospheric and crustal flow powered by slab retreat east of the collision zone below East and SE Asia. The main shortening direction in the deforming continent between the collision zone and the Pacific subduction zones may in this case be a proxy for the direction of flow in the asthenosphere underneath, which may become a useful tool for studying mantle flow in the distant past. Our model of the India-Asia collision emphasizes the role of asthenospheric flow underneath continents and may offer alternative ways of understanding tectonic processes.

**Plain Language Summary** The collision of India with Asia has induced the propagation of continent-scale faults to large distances from the Himalayas, all the way to the margins of Asia where back-arc basins, such as the Japan Sea, formed above the subductions of the Pacific and Indian Oceans. The respective roles of collision and subduction in controlling the deformation of Asia at such long distances are not well understood. Based on a comparison of kinematic trajectories of lithospheric plates and blocks since 50 Ma with mantle flow directions inferred from seismological data, we propose that the connection between collision and back-arc extension is made by the asthenosphere flowing underneath. This flowing mantle comes from an upwelling zone below South Africa and has pushed India inside Asia along more than 3,000 km after the first collision. The flow then invaded Asia all the way to the subduction trenches where it followed the retreat of the subducting lithospheres, controlling the geometry and kinematics of the major strike-slip faults and back-arc basins. This possible control of continental deformation by the asthenosphere flowing underneath is a different view of continental deformation and may offer a way of studying mantle flow below continents in the distant past.

## **1. Introduction**

Since the early days of the continental drift theory, the question of the engine of surface displacements and formation of rifts or mountain belts has been discussed in terms of mantle convection (Griggs, 1939; Holmes, 1931; McKenzie, 1969; Runcorn, 1962; Wilson, 1973). More recent models have shown that mantle flow and slab suction are important ingredients to explain plate kinematics (Conrad & Lithgow-Bertelloni, 2002) but deformation within plates is rarely considered in these models (Dal Zilio et al., 2017; Faccenna, Becker, Conrad, et al., 2013; Faccenna, Becker, Jolivet, et al., 2013; Jolivet et al., 2009; Yamato et al., 2013). Yet we know that different styles of convection may have significant consequences for the dynamics of mountain belts. For

instance, whether slabs enter the lower mantle or not leads to orogens controlled by slab-pull or slab suction and the transition between the two styles of subduction can lead to large-scale obduction (Faccenna, Becker, Conrad, et al., 2013; Faccenna, Becker, Jolivet, et al., 2013; Husson et al., 2012; Jolivet et al., 2016). Although it was shown that the mantle flow induced by slab retreat is able to impose shearing conditions at the base of the overriding plate and controls deformation in the back-arc region (Capitanio, 2014; Capitanio et al., 2015; Chen et al., 2016; Faccenna et al., 2006; Jolivet et al., 2009; Menant, Sternai, et al., 2016; Sternai et al., 2014), at larger scale, few tectonic models explain crustal deformation considering a possible drag by underlying currents (Conrad & Lithgow-Bertelloni, 2006; Ghosh et al., 2013; Koptev et al., 2015, 2016; Stoddard & Abbott, 1996). The simple observation that the India-Asia collision has not stopped the northward motion of India, despite the absence of a well-defined deep slab below Tibet, demonstrates that other forces than slab-pull should be considered (Alvarez, 2010; Becker & Faccenna, 2011; Becker & O'Connell, 2001; Faccenna, Becker, Conrad, et al., 2013; Warners-Ruckstuhl et al., 2010).

The case of the deformation of East Asia is particularly significant in this regard. Since the collision with India, the Asian lithosphere has been deforming across a wide region from the Himalaya to the Pacific and Indian Oceans subduction zones (Jolivet et al., 1994; Kimura & Tamaki, 1986; Royden et al., 2008; Tapponnier et al., 1982, 2001). Crustal shortening in the vicinity of the collision zone gave rise to the Himalaya and the Tibetan Plateau, but deformation is observed at large distance in the form of large-scale strike-shear zones and faults, controlling the opening of rifts (e.g., Baikal rift) and back-arc basins (Japan Sea, South China Sea; Jolivet et al., 1994; Kimura & Tamaki, 1986; Tapponnier et al., 1982, 2001). The dynamic interactions between collision and back-arc opening, and thus subduction, have never received a proper explanation, since most models consider the eastern margin of Asia as a simple free boundary.

Here we show that the tectonic history of eastern Asia can be explained if the flow of asthenospheric mantle carrying India northward reaches the eastern and southeastern margins of Asia and interacts with the Pacific and Indian slabs. Based on a comparison of long-term kinematic trajectories, inferred stress field and seismic anisotropy, we show that the large-scale deformation of Asia can be explained by the coeval continent-continent collision in the Himalaya-Tibet region and slab retreat east and southeast of Asia. These two regions are connected by the asthenospheric mantle flowing underneath, from the collision zone to back-arc regions.

## 2. Geodynamic Setting

The far-field deformation of Asia during the collision with India (Figure 1) and more specifically the relations between shortening in the Himalaya and opening of back-arc basins at the periphery of the system (Jolivet et al., 1994; Tapponnier et al., 1982) poses the question about the nature of the interactions between these two distant convergence zones. Compressional deformation naturally characterizes the region close to the India-Asia collision zone (Himalaya and Tibet; Royden et al., 2008; Tapponnier et al., 2001), and gravitational forces stored in the thick crust of Tibet largely control the kinematics of convergence since the time of collision (Copley et al., 2010). On the other hand, large-scale strike-slip faulting (Molnar & Tapponnier, 1975) and extension are characteristics of a wide domain encompassing most of eastern Asia (Jolivet et al., 1994; Tapponnier et al., 1982; Yin, 2010; Figures 1 and 2). Furthermore, deformation related to collision is asymmetrically distributed and shifted eastward toward the Pacific margin (Davy & Cobbold, 1988; Dewey et al., 1989; Tapponnier et al., 1982). Previous studies have often not considered that

- i A large part of East and Southeast Asia rests on a thin and hot lithosphere that extends westward of the western Pacific trench and northward of the Sunda trench, as far as the Mongolian Plateau where Cenozoic volcanism has been widespread since 30 Ma (Hunt et al., 2012; Smith et al., 2016; Steinberger & Becker, 2016; Tesauro et al., 2013; Windley & Allen, 1993; Yu et al., 2017).
- ii Back-arc extension above the Pacific subduction zone has been active between at least 50 and 10 Ma (Figure 2), interfering with collision-related deformation during this period (Fournier et al., 2004; Jolivet et al., 1990, 1994). Back-arc extension above the Sunda subduction zone started earlier in the Paleogene forming a series of distributed rifts and oceanic basins, including the largest of them, the South China Sea (Hall, 2002; Pubellier & Morley, 2014; Rangin et al., 1990) that formed in the same period as the Japan Sea.
- iii Back-arc basins have formed within the Philippine Sea Plate (PHSP) since the Paleogene until the Present during its northward movement, forming the West Philippine Basin, the Shikoku-Parece Vela Basin, and the Mariana Basin (Chamot-Rooke et al., 1987; Hall, 2002; Lallemand, 2016).





**Figure 1.** (a) The main tectonic features of the Cenozoic Eurasia. The map also shows the extent of a high *P*-velocity anomaly in the lower mantle (1,100 km) below India and Southeast Asia (Hall & Spakman, 2015; Replumaz et al., 2004). This anomaly shows the position of the subducted Tethyan slab and the difference between India that has overpassed its slab and Southeast Asia where slab retreat has occurred. This figure shows that the whole eastern part of Asia has been extended, leading to extreme lithospheric thinning and formation of oceanic crust in western Pacific back-arc basins (Jolivet et al., 1999). AB = Anatolian Block; AS-RV SZ = Ailao Shan-Red River Shear Zone; ATF = Altyn Tagh Fault; BR = Baikal Rift; EAR = East African Rift; IBT = Izu-Bonin Trench; JS = Japan Sea; JT = Japan Trench; MT = Mariana Trench; OB = Okinawa Basin; PVB = Parece Vela Basin; SB = Shikoku Basin; SCS = South China Sea; SH SZ = Sakhalin-Hokkaido Shear Zone; ST = Sunda Trench; WPB = West Philippine Basin. The white arrow shows the point where the deep mantle velocity anomaly (Tethyan slab) crosses the Sunda Trench indicating the transition along strike from advancing to retreating subduction. (b) Lithospheric thickness map after Tesauro et al. (2013) and the main tectonic features of the Tethys convergence zone. This map illustrates the thinning of the lithosphere below East Asia. White arrows show the escape of continental blocks with opposite sense in the eastern Mediterranean region and eastern Asia.





**Figure 2.** Tectonic maps of Asia and the western Pacific. (a) Present-day; (b) a schematic reconstruction at 25 Ma. This reconstruction is based on Torsvik and Cocks (2016) for the motion of large plates with modifications in back-arc regions (Barckhausen et al., 2014; Jolivet et al., 1989, 1994; Pubellier & Morley, 2014; Rangin et al., 1999, 2013). AS RV SZ = Ailao Shan-Red River Shear Zone; LMS = Long Men Shan; OB = Okinawa Basin; SH SZ = Sakhalin-Hokkaido Shear Zone; SPVB = Shikoku-Parece Vela Basin; WPB = West Philippine Basin.

The progressive collision process can be summarized as follows: at ~50 Ma, the Neo-Tethys Ocean closed and the Indian continental lithosphere started to subduct (Bouilhol et al., 2013; van Hinsbergen et al., 2011). The geometry of the initial contact is unknown, but the subduction zone could have been rather straight if considering the present-day shape of the high velocity anomaly in the lower mantle representing the lost Tethys Ocean (Figure 1a; Replumaz et al., 2004). Since that time, India has progressed northward more than 3,000 km and the Himalayas and the Tibetan Plateau have progressively developed above the continental subduction zone (Royden et al., 2008; Tapponnier et al., 2001). Within the Tibetan Plateau, active deformation is currently partitioned between strike-slip faulting and E-W extension (Molnar & Lyon-Caen, 1989; Royden et al., 2008; Tapponnier et al., 2001). Furthermore, earthquake focal mechanisms and the GPS velocity field show an extrusion of crustal material toward the east and southeast with a progressive decrease of velocities (Molnar & Lyon-Caen, 1989; Zhang et al., 2004). How far this pattern can be traced back in time is unknown, but extension started at least between 13 and 18 Ma ago (Coleman & Hodges, 1995; Williams et al., 2001). This timing shows that since the Middle Miocene at least, body forces stored in the thick crust of Tibet have been competing with the boundary forces originating from India-Asia convergence and that the plateau was already at high elevation. Thermochronological studies suggest that the growth of the plateau started early, since 35 Ma or earlier (Rohrmann et al., 2012; van der Beek et al., 2009). The plateau has indeed formed progressively with constant denudation rates since at least 35 Ma, only 15 Ma after the first collision or even since the Late Cretaceous.

The 25 Ma schematic reconstruction of Figure 2b is based on Torsvik and Cocks (2016) and modified for backarc basins using Jolivet et al. (1989, 1994), Rangin et al. (1990), Pubellier and Morley (2014), Hall, Ali, and Anderson (1995), Hall, Ali, Anderson, and Baker (1995), and Hall (2002). Outside the collision zone, the deformation of the Asian lithosphere is less distributed and large-scale strike-slip faults have formed, such as the Ailao-Shan Red River Shear Zone (Figures 1 and 2), accommodating the eastward escape of continental blocks, such as Indochina. Although the exact amount of left-lateral displacement along the Ailao Shan-Red River Shear Zone is a point of debate and is not known with precision (from 100 to 1,000 km; Dewey et al., 1989; Huchon et al., 1994; Tapponnier et al., 1982, 1990), these faults are the most prominent structures seen outside the Tibetan Plateau, apart from those of the Baikal Rift and Tien Shan Range. These faults connect the compressional domain in the collision zone with the extensional domain along the Pacific margin. The India-Asia collision thus induced the formation of large-scale lithospheric faults that diverge from the contact with the indenter (Molnar & Tapponnier, 1975; Tapponnier & Molnar, 1977), and consequently the displacement of lithospheric blocks eastward and south-eastward. During the same period, a series of large dextral shear zones, such as the Sakhalin-Hokkaido Shear Zone (Figures 1 and 2), formed in East Asia, conjugate of the left-lateral ones. These dextral faults interfered with the opening of back-arc basins, such as the Japan and Okhotsk Sea or the Bohai Basin (Fournier et al., 2004; Jolivet et al., 1990, 1991, 1994; Lallemand & Jolivet, 1985; Yin, 2010). The Sakhalin-Hokkaido Shear Zone (Jolivet, 1986; Jolivet et al., 1994; Lallemand & Jolivet, 1985) forms the western boundary of the relatively rigid Okhotsk block, which has been active under transpressional dextral shear since the Oligocene. It has also been interpreted as the active boundary between the North American and Eurasia plates south of the rotation pole located in East Siberia (Chapman & Solomon, 1976). Since the present paper concerns mainly the Oligo-Miocene period, we will use the name Sakhalin-Hokkaido Shear Zone in the following.

Continent-continent collision induces compressional stresses that diverge from the collision zone, transmit through the rigid lithosphere, and progressively decrease away toward the extending eastern boundaries of Asia. This asymmetrical distribution of deformation is classically explained by the concept of a free boundary along the eastern margin of Asia, which allows the motion of lithospheric blocks accommodating the indentation of India into Asia (Jolivet et al., 1990; Kimura & Tamaki, 1986; Tapponnier et al., 1982). The general geometry is compatible with the different models proposed so far, rigid-plastic indentation and slip-line field theory (Molnar & Tapponnier, 1975; Tapponnier & Molnar, 1976) or propagating extrusion models (Tapponnier et al., 1982). Analogue indentation models scaled for gravity (Cobbold & Davy, 1988; Davy & Cobbold, 1988; Jolivet et al., 1990) indeed induce the formation of a conjugate set of sinistral and dextral strike-slip faults that connect the collision zone with the free boundary to the east. When adding a component of extension using a gravitationally unstable lithosphere that spreads under its own weight (Fournier et al., 2004), the deformation is less localized, but the overall geometry is preserved, with the addition of a wide domain of distributed extension in Southeast Asia. However, in these models, there is no subduction

east of the collision zone and they do not account for any possible mantle flow contribution to the deformation.

In fact, the extensional component in Southeast and Northeast Asia is driven by slab retreat and return flow east of the collision zone (Jolivet et al., 1994; Sternai et al., 2016). Based on analogue modeling, Schellart et al. (2003) proposed that noncylindrical rollback of the subduction zone can explain the link between back-arc rifting and the formation of the dextral shear zone separating the Okhotsk and Japan Sea, but in this case no link with the larger-scale deformation of Asia was taken into account. Yin (2010) proposed that the dextral component observed during back-arc rifting was due to the obliquity of the Pacific subduction but, again, no link with collision was made. The interactions between subduction dynamics and far-field effects of the India-Asia collision thus remain poorly understood.

It has also been shown that the flowing asthenospheric mantle, beside a possible drag effect through basal shear on large surfaces (Bird, 1998), can generate a force in presence of topographic irregularities of the lithosphere-asthenosphere boundary (LAB), which induces deformation in the continent above (Conrad & Lithgow-Bertelloni, 2006; Ghosh et al., 2013; Koptev et al., 2015; Stoddard & Abbott, 1996). We now explore this hypothesis for the case of Cenozoic Asia and the mantle flowing underneath, until the peri-Pacific sub-duction zones, through the comparison of the long-term (50 Ma) kinematics of the convergence zone, lithospheric thickness, mantle fabric (seismic anisotropy), and slab geometries.

We assume that long-term kinematic trajectories reveal the first order flow of lithospheric material and that seismic anisotropy is a proxy for the mantle strain, and thus flow. We compare kinematic trajectories with different sets of seismic anisotropy constraints (SKS-splitting and surface waves) and discuss the observed similarities and differences. We use the geometry of slabs, from the Himalayan slab to the Pacific and Indian slabs, to infer the direction of the mantle flow. Based on the last observations, we propose a scenario that fits the succession of the tectonic events and the stress field data and discuss the implications in terms of coupling between mantle flow and crustal deformation.

## 3. Kinematic Trajectories

We calculate long-term (50 Ma) kinematic trajectories with GPlates (Boyden et al., 2011) based on the absolute model of Torsvik and Cocks (2016), the Global Hybrid Mantle Reference Frame (GHMRF), which is based on the absolute motion model (moving hot spots) of Doubrovine et al. (2012) and the palaeomagnetic reference frame corrected for true polar wander (Torsvik et al., 2012; Figure 3a). These trajectories show a rather simple long-term motion, with India and Africa moving NNE and a progressive clockwise rotation of the flow toward the eastern margin of Asia, where it meets the westward Pacific flow. On such a long time period, one may assume that the effects of first order, large-scale processes (several thousands of kilometers) on mantle fabrics dominate over more local kinematic changes. For instance, the recent (~10 Ma) change of stress regime around Japan from extension to compression (Becker et al., 2006; Ingle, 1992; Jolivet et al., 1994) has been associated with small finite displacement and is not representative of the long-term tectonic evolution of this region characterized by back-arc rifting and spreading. The choice of 50 Ma as a starting point is based on two main considerations: We want to span a period long enough to have a stable fabric in the mantle (Becker et al., 2006) and we know that plate motions have not significantly changed since the time of the abrupt change in the kinematics of the Pacific Plate between 53 and 47 Ma (Seton et al., 2015).

We first explored the effects of several plate motion reference frames on the kinematic trajectories (Figures 3 and S1). We modified the model by Seton et al. (2012) and Zahirovic et al. (2015) to incorporate local reconstruction features in the Mediterranean region (Jolivet et al., 2003; Menant, Jolivet, & Vrielynck, 2016) and eastern Asia (Jolivet et al., 1991, 1994). Trajectories between 50 Ma and the present were calculated in three reference frames, fixed Eurasia, fixed Atlantic and Indian Ocean hot spots, and fixed Antarctica. Two models are in an absolute reference frame, with either a mantle reference (Atlantic and Indian Oceans hot spots fixed; Seton et al., 2012; Zahirovic et al., 2015) or the GHMRF reference frame (Torsvik & Cocks, 2016; Figure S1). Models in the absolute reference frames show similar patterns with a dominant component of northward motion of Africa and India, an eastward motion of Eurasia and westward motion of the Pacific Plate. These models differ mostly for slow plates (Eurasia and Africa). A reasonable fit with seismic anisotropy is obtained with the mantle-fixed frameworks, and we finally chose the GHMRF model (Figure 3b). As we will discuss in



**Figure 3.** (a) Kinematic trajectories since 50 Ma (one step every 5 Ma, from red to blue) on top of the S-wave velocity anomalies of the lower mantle (LLSVP at 2,800 km) after Becker and Boschi (2002). (b) Lithospheric thickness map after Steinberger and Becker (2016) and kinematic trajectories since 50 Ma in an absolute kinematic reference frame based on paleomagnetism, calculated after Torsvik and Cocks (2016). The same figure also shows the smoothed distribution of SKS fast splitting directions on top of lithospheric thickness. The color code of smoothed SKS fast splitting direction indicate the angle with the long-term kinematic trajectories (green 0–20°, blue 20–40°, red above 40°). This figure shows that both asthenospheric flow (through the seismic anisotropy proxy) and kinematic trajectories suggest a mantle flow diverging away from the Tuzo LLSVP with a main direction toward the north and northeast.

section 5, the best match is observed below northeast Asia that is supported by a thin lithosphere and below India that is moving fast, driven by the northward mantle flow.

The trajectories that we obtained provide an image of the flux of crust and mantle lithosphere since 50 Ma. Figure 3a shows these trajectories plotted over a map of seismic velocity anomalies in the lower mantle (2,800 km) from the composite model of Becker and Boschi (2002). It shows the two large low shear velocity provinces below West and South Africa (*Tuzo* in Burke, 2011) and below the Pacific Ocean (*Jason*) and the high velocity anomaly in between, below the West Pacific subduction zones. The long-term trajectories of



Africa, India, and Eurasia show a flow away from Tuzo toward the West Pacific subduction zones, suggesting a simple connection with the large-scale mantle convection (i.e., the *conveyor belt*; Becker & Faccenna, 2011).

## 4. Lithospheric Thickness

The pattern of lithospheric thickness is a stable feature for several methods used, as shown by the comparison of two models (Steinberger & Becker, 2016; Tesauro et al., 2013; Figures 1b and 3b). In a first approach (Tesauro et al., 2013), the temperature of the lithosphere on a global scale was estimated from the inversion of a recent seismic tomography model (Ritsema et al., 2011), using a mineral-physics approach (Cammarano et al., 2003). This approach consists of calculating the anharmonic velocities at given pressure and temperature conditions by estimating density and the anharmonic moduli, first in temperature and then, along an adiabat, in pressure. The composition used is representative for the mantle lithosphere over large areas, such as Proton-Archon in the cratons (Griffin et al., 2003), which is depleted in iron content, and PrimitiveMantle (McDonough & Sun, 1995) outside the craton. The effect of anelasticity was taken into account using the homologous temperature scaling approach, since large uncertainties exist in estimates of activation enthalpy (Karato, 1993). The depth of the thermal LAB displayed in Figure 1b has been estimated from a global thermal model (Tesauro et al., 2013), as the depth of the 1200 °C isotherm, intermediate between the 1300 °C (mantle solidus temperature) and 1100 °C (0.85 of the mantle solidus temperature) isotherm (Artemieva & Mooney, 2001). Such a choice is justified by the fact that mechanical properties of the mantle may change gradually in the vicinity of the solidus. Therefore, no sharp boundary between the mechanically weak lithosphere and the asthenosphere (LAB) possibly exists (Cammarano et al., 2003).

The second lithospheric thickness model (Steinberger & Becker, 2016; Figure 3b) is based on the depthvariation of seismic velocity anomalies from tomographic models. A reference, half-space cooling type of depth dependence is assumed to control the thermal contribution to velocity anomalies, and then an effective thermal lithospheric thickness is derived after correction for a compositional contribution, which is of particular importance in the continents. Steinberger and Becker (2016) explore the uncertainties involved in this approach and compare thickness estimates from their approach with those obtained using other constraints. Here we use their mean model of lithospheric thickness, which is based on averaging the estimates from several recent, global shear wave tomography models, with a correction for mismapped thickness due to Wadati-Benioff zones. In these two maps (Figures 1b and 3b), thickness gradients are different but the main features are preserved in both models.

The main observation from these maps is that the extended and thin lithosphere of East and Southeast Asia lies over the leading edge of the southeastward lithospheric flow (Figures 3 and 4). This situation is reminiscent of the evolution of the African Plate since the Paleozoic with the rifting and detachment above the northward mantle flow of continental ribbons that then crossed the Tethys Ocean until they finally collided with the southern margin of Eurasia (Jolivet et al., 2016).

### 5. Seismic Anisotropy

In order to compare long-term displacement trajectories with inferred mantle flow, we use seismic anisotropy as a proxy for mantle deformation, assuming that mantle peridotites are stretched parallel to shearing direction. Azimuthal seismic anisotropy refers to the dependence of seismic velocity on the azimuth of wave propagation, a property acquired by rocks through their deformation. Different ways of measuring this anisotropy are available and we use two of them in this paper. (1) The widely used SKS-splitting method uses local measurements of the direction and delay of the maximum velocity difference between polarized components of a seismic wave traveling in an anisotropic medium (Savage, 1999). (2) Anisotropy can also be extracted from global tomographic models; here we use a recent global shear speed anisotropy model (Schaeffer et al., 2016; Schaeffer & Lebedev, 2013), SL2013sv hereafter. Rather than giving a direct image of mantle flow, seismic anisotropy is rather a proxy for deformation as the signal depends upon rock fabrics (e.g., Long & Becker, 2010). We will assume that the deformation in the mantle is close to plane strain and that the flow is close to horizontal. Neglecting vertical deformation, the azimuth of the fast direction is then parallel to the shearing direction in map view.

Comparing surface (crustal) deformation and seismic anisotropy in the mantle (Figures 3b and 4) allows discussing the degree of coupling between crustal and mantle strain pattern (Chen et al., 2016; Faccenna et al.,





**Figure 4.** SL2013sv azimuthal anisotropy (colored bars) from Schaeffer and Lebedev (2013) at 100 (a) and 200 km (b) depth. These plots show a distinct change between 100 and 200 km. Green, blue, and red colors show the quality of the fit (from best to poorest, same code as in Figure 3) of seismic anisotropy and kinematic trajectories. At 100 km, the fast velocity pattern shows close correspondence with crustal features and a north-south direction predominates in Southeast Asia, compatible with the expected direction of slab retreat behind the Java Trench in the low-velocity domain. At 200 km, the whole of East and Southeast Asia shows a SW-NE main direction suggesting a continuation of the flow carrying India northward. Inset in Figure 3a shows the velocity field obtained in a numerical model of continent-continent collision (Sternai et al., 2016).

2014; Faccenna & Becker, 2010; Flesch et al., 2005; Jolivet et al., 2009; Little et al., 2002; Tikoff et al., 2004). The SL2013sv anisotropy model allows visualizing the stratification of azimuthal anisotropy, while SKS-waves show more local anisotropy averaged over the whole thickness of the mantle below the station. Most authors consider that seismic anisotropy is due to shearing between the moving rigid lithosphere and the viscous asthenosphere (Flesch et al., 2005; Lavé et al., 1996), assuming that the asthenosphere responds passively. For instance, Flesch et al. (2005) or Wang et al. (2008) show a vertically coherent deformation below Tibet, from the crust to the mantle, and they conclude that the lithospheric mantle has survived

orogenic processes as it can transmit crustal buoyancy forces downward. If one instead assumes that part of the deformation is driven from below by the viscous asthenosphere, the presence of a passively deforming mantle is no longer required. More detailed approaches of the stratification of anisotropy below Tibet show a different behavior at lithospheric and asthenospheric depths interpreted as different flow directions (Agius & Lebedev, 2017; Huang et al., 2017).

We used two sets of azimuthal seismic anisotropy observations (Figures 3b and 4), (1) a compilation of SKS-splitting anisotropy (Becker et al., 2012) and (2) the tomographic model SL2013sv (Schaeffer et al., 2016; Schaeffer & Lebedev, 2013). We further compared SL2013sv with the model of Debayle et al. (2016) and found that differences in azimuthal anisotropy are overall moderate, at least in terms of directions, implying that the observed patterns are robust.

The patterns of SKS-splitting anisotropy and SL2013sv (Figures 3b and 4) are different, because the resolutions of the two methods are different and likely also because they sample different processes and different depths (e.g., Becker et al., 2012). However, some similarities can be observed when comparing smoothed SKS fast splitting orientations and SL2013sv. We assess the fit between the different observations by measuring the angle between the orientations and indicate the range of angles with different colors of symbols (green for angles between 0° and 20°, blue for angles between 20° and 40°, and red for angles greater than 40°). The 100-km depth anisotropy map for SL2013sv (Figure 4a) shows a good match with SKS below continental regions with thin lithosphere (Figure S3). In particular, there is a good fit in the Mediterranean, Middle East, part of North and East Africa, part of India, and Northeast Asia, while the fit is reasonable below Tibet. At 200 km (Figure 4b), the Mediterranean region, North and east Africa, and part of India still show a good to fair fit (Figure S3), but Tibet and Northeast Asia show very different trends. Southeast Asia shows a poor match at both depth ranges. When considering a depth of 25 km below the LAB (Figures 5b and S4–S6), the two models of lithospheric thickness show a similar fit of SL2013sv anisotropy and SKS-splitting direction to first order. The northeastern part of Africa, Arabia, India, northeast Asia, and part of the Indian Ocean and Tibetan Plateau show a good or fair fit.

The best fit of SL2013sv anisotropy and SKS fast splitting orientations is thus below regions of thin lithosphere and at shallow depth, either 100 or 25 km below LAB (Figures 5 and S3). One notable exception is below SE Asia where the two sets of data give perpendicular directions, except at 200 km.

## 6. Comparing Seismic Anisotropy and Long-Term Kinematic Trajectories

We now compare seismic anisotropy and long-term kinematic trajectories, which shows different results when using either SKS or SL2013sv. For SKS-splitting anisotropy (Figure 3b), the fit is good (green symbols) or reasonable (blue symbols) below India and part of the Indian Ocean, as well as below regions of thin lithosphere, such as East and part of North Africa, from Libya to the East African Rift, and below a large part of East Asia, especially between Japan and the Baikal Rift and Hangay dome. The fit is however poor (red symbols) below Southeast Asia and the central and West Mediterranean realm. A more recent investigation over Indochina (Yu et al., 2018) confirms this poor fit. When using SL2013sv's seismic anisotropy inference instead (Figure 4), the quality of the match depends upon the depth considered. At 100 km (Figure 4a), it is good over a large part of the Indian Ocean, south of India, below the Tibetan Plateau, and most of Northeast and Southeast Asia. This good agreement disappears at 200 km (Figure 4b) below Northeast and Southeast Asia. At this depth, the fast direction in SL2013sv trends NE-SW below India and the Indian Ocean and slowly bends toward E-W below China and Japan. Another important observation is that SL2013sv azimuthal anisotropy at 200 km is very weak below Indonesia, especially north of the E-W trending part of the Sunda Trench. This can either be due to an intrinsically weak anisotropy of the mantle or to steeply dipping stretching in the mantle.

In regions where both SKS- and surface-wave anisotropy provide the same sense of mantle shear (Figure S7), this orientation is also parallel to the long-term trajectories in an absolute kinematic framework, which is in agreement with the assumptions made above (horizontal flow and plane strain). In this case, strain and flow are coaxial in map view. However, the case of the Mediterranean region is different, as the trajectories in an absolute framework are highly oblique to the fast directions in the mantle (Figure S2). There, surface kinematics are dominated by the retreat of narrow slabs that move fast in the mantle (Faccenna et al., 2004;





**Figure 5.** Comparison of SL2013sv anisotropy directions (colored symbols) with long-term kinematic trajectories (40 Ma, white lines) calculated from Torsvik and Cocks (2016) at 100 km (a) on top of a map of integrated strength of the lithosphere, based on Tesauro et al. (2013), and (b) at 25 km below LAB in the model of Tesauro et al. (2013), the background is the lithospheric thickness model of Tesauro et al. (2013).

Spakman & Wortel, 2004), with velocities well above 10 cm/year. So, in that case too, seismic anisotropy may give a good image of the flow.

In the case of East Asia, the vertical superposition of two perpendicular stretching/shearing directions, seen here at large scale, is confirmed by detailed studies of the stratification of anisotropy below Tibet (Agius & Lebedev, 2017; Huang et al., 2017) with different flow directions at lithospheric and asthenospheric depths. One can reasonably assume that the signal at 200 km for SL2013sv gives information on the asthenospheric flow as it samples the mantle well below the LAB in East Asia. At 100 km, the situation is more complex as the map of SL2013sv anisotropy probably contains some signal provided by the lithospheric mantle or even the lower crust, as proposed for Tibet (Agius & Lebedev, 2017). The 100-km map should then be considered as an image of the flow/strain in the lithosphere and the 200 km map as the flow in the asthenosphere.



## 7. Discussion

Mantle stretching appears in general well correlated with long-term kinematic trajectories in regions of thin lithosphere. These include the eastern Mediterranean, East Africa, and Northeast and Southeast Asia at 100 km (for SL2013sv only). We discuss in the next sections the implications of these observations at the scale of Eurasia with a focus on East Asia.

#### 7.1. Deep Flow Pattern

The best fit of seismic anisotropy and kinematic trajectories is thus observed at the depth of 100 km for regions supported by a lithosphere thinner than 100 km, but also beneath Tibet where the lithosphere is thicker (Figures 3b and 4a). There, the LAB is shallower than 100 km, but the crust is very thick, which also confirms a link between thin (and thus hot) lithospheric mantle and a good coupling between surface displacements and mantle deformation. This is confirmed by plotting (Figure 5) SL2013sv anisotropy at 100 km or SKS fast splitting orientations on a map of the integrated strength of the lithosphere (Tesauro et al., 2013), showing that the whole of East Asia rests on a weak lithosphere, including the thick crust of the Tibetan Plateau. Therefore, a weak lithosphere favors coupling between surface displacements and mantle deformation more than its absolute thickness. At 200 km (Figure 4b), a good fit is observed below regions where fast northward mantle flow can be inferred, such as below India and a large part of the Indian Ocean, or below East Africa where the Afar plume drives the opening of the Red Sea and the drift of Arabia (Faccenna, Becker, Jolivet et al., 2013). Below East Africa, the match is more in agreement with SKS-waves anisotropy than with that obtained from the surface wave model.

In Southeast Asia, SKS-wave anisotropy does not match the plate trajectories (Figure 3b), while SL2013sv anisotropy instead shows a good fit around 100 km (Figure 4a). This may imply that SKS-splitting anisotropy, as it corresponds to local measurements, shows here a smaller-scale flow that is not captured by the smooth SL2013sv model. The direction of this flow could be toward the west or southwest, corresponding to a tighter counter flow around the East Himalayan syntaxis toward the Burma slab as proposed by Wang et al. (2013) and Yu et al. (2018). At 100 km, the flow geometry (or of shearing below the lithosphere) indicated by SL2013sv anisotropy below Asia can otherwise be compared with the model of Sternai et al. (2016; Figure 4a, inset) which was designed to study the 3-D interactions between slab retreat and collision for the India-Asia collision. This model suggests that the mantle flowing underneath East Asia exerts a control on the tectonic evolution and the topography of the domain situated to the east of the east Himalayan syntaxis. The three plots (insets in Figure 4a) showing the upper crustal, lower crustal, and asthenospheric flows illustrate this similarity. The asthenospheric flow is here entirely driven by convergence and slab retreat (there is no larger-scale flow involved in the model as the plume-related deeper flow carrying India is not considered). The observed similarity suggests that the pattern of SL2013sv anisotropy observed at 100 km is a consequence of the India-Asia collision and slab retreat south of Indonesia and not of the deeper large-scale conveyor belt.

The main question is then the cause of this widely distributed extension across eastern and southeastern Asia. If asthenospheric flow plays a role in the distribution of deformation across Asia, the geometry of surrounding slabs should reflect it. We then now consider the geometry of slabs as inferred from seismic tomography models.

#### 7.2. Slabs Geometries

Figures 6 and 7 show a synthesis of the shapes of slabs around Asia based on seismic tomography (Amaru, 2007; Li et al., 2008; Obayashi et al., 2013; Replumaz et al., 2004; Simmons et al., 2012; van der Meer et al., 2010). Subducted lithosphere shows an opposite behavior when comparing the Himalaya-Tibet region and the subduction zones surrounding Asia to the east and southeast.

#### 7.2.1. Indian Slab below the Himalayas and Tibet

The Indian slab is discontinuous and oversteepened (Figure 6a). The upper part of the slab (white arrow 1 in Figure 6a) is still attached to the Indian lithosphere and does not go further down than the upper mantle. Two high velocity anomalies (white arrows #2 an #3 on Figure 6a) in the lower mantle reflect the multistage India-Eurasia convergence and/or indicate that the slab has detached twice (van der Meer et al., 2010; Van der Voo et al., 1999), leaving Tethyan slab remnants behind, which have continued to sink in the lower mantle (Replumaz et al., 2004, 2013). India has thus overridden the broken-off Tethyan slab remnants during



**Figure 6.** The shapes of slabs around Asia. The upper left panel shows the detached and overhanging Himalayan slab (van der Meer et al., 2010). Lower panels show two sections across the Japan subduction zone with four different models: UU-PO7 (Amaru, 2007), GAP-P4 (Obayashi et al., 2013), MIT-2008 (Li et al., 2008), and LLNL\_G3Dv3 (Simmons et al., 2012). Upper right panel: Three time frames of the evolution of West Pacific and East Asia showing the subduction of the Pacific-Izanagi ridge (Seton et al., 2012; Zahirovic et al., 2015). Part of figure was prepared with GPlates (Boyden et al., 2011).

collision, advancing ~3,000 km northward inside Asia since 50 Ma. At present, part of the Indian slab lies horizontally below the eastern part of Tibet (Agius & Lebedev, 2013), indicating that the last break-off event resulted in slab rebound, a process that implies an external force pushing the Indian slab northward (Magni et al., 2017). The geometry of the Tethyan slab now seen in the lower mantle (Hall & Spakman, 2015; Replumaz et al., 2004; Figure 1a) shows the transition between the advancing and retreating foreland basins/trenches from the Himalayas (advancing) to the Sunda Trench (retreating), where its projection at the surface crosses the subduction zone (white arrow on Figure 1a).

#### 7.2.2. Subducting Slabs in Sunda and Peri-Pacific Trenches

On the eastern edge of Asia, along the western Pacific margin, and in contrast to the Tethyan realm, the oceanic slabs subducting below Asia to the east and southeast instead show concave-up geometry in the upper mantle (Figure 6b). By *concave-up* we mean that the shallow slab dips steeper than the deeper slab that lies horizontally at the bottom of the transition zone. This geometry is for instance observed at the latitude of





**Figure 7.** Sketch of the displacement and deformation of the Himalayan and Pacific slabs by the flow of Indian mantle below Asia and mantle flow at asthenospheric depth (black arrows) and lithospheric depth (white arrows). Small green arrows show the recent velocity field on the Tibetan Plateau. Large green arrows on the upper sketch shows the main stretching direction. The asthenospheric flow has carried India northward inside Asia. The geometry of slabs was sketched based on seismic tomography (Agius & Lebedev, 2013; Amaru, 2007; Li et al., 2008; Obayashi et al., 2013; Replumaz et al., 2004; Simmons et al., 2012; van der Meer et al., 2010; Zhao et al., 2017). The flow carries also the slab observed below the Himalaya and it extends behind the collision zone all the way to the Pacific margins. The indentation of India into Asia has formed the Himalayan Range and the Tibetan Plateau in the compressional zone near the collision front. In the east, the leading edges of Asia are subjected to extension above eastward flowing asthenosphere following slab retreat. This extensional stress field in the east makes the boundary conditions asymmetrical and the deformation of Asia due to collision thus develops in an asymmetric way with the main strike-slip faults developing toward the east. The lateral transition from continental collision in the Himalayas to oceanic subduction in the Sunda Trench leads to a more superficial return flow that drags the edges of Asia southward toward the retreating Sunda Trench. Inset shows a map of Eurasia with the main flow direction associated with Arabia-Eurasia and India-Asia collisions with respect to Eurasia.

Japan and Ryukyu Islands, as illustrated by different tomographic data sets (Amaru, 2007; Fukao & Obayashi, 2013; Li et al., 2008; Obayashi et al., 2013; Pownall et al., 2017; Simmons et al., 2012; Zhao, 2017; Zhao et al., 2017). Goes et al. (2017), based on a review of the geometry of slabs worldwide and of dynamic models,

review that this geometry is due to both (1) the viscosity increase and the phase transition between the upper and lower mantle and (2) trench retreat.

The age of the subducted oceanic lithosphere east of Asia has been so far diversely interpreted. Honda (2016) proposed that the edge of the slab stagnating below Northeast China is the subducted mid-ocean ridge between Izanagi and Pacific Plates. However, this proposition bears an inconsistency: since 50 Ma, some 5,000 km of Pacific lithosphere have subducted toward the west below Asia at the latitude of Japan, which is much more than the observed length of the subducted lithosphere between the Japan Trench and the tip of the horizontal slab at depth (~1,500 km only). A more recent analysis (Liu et al., 2017) suggests instead that the slab seen today in the upper mantle started to subduct 25–30 Ma ago and that the tip of the flat slab is 90 Ma old, while the oceanic lithosphere entering the trench today is 130 Ma old. This alternative interpretation has several implications: (1) The slab was already old and dense enough to drive slab retreat in the Oligocene and Miocene, (2) the flat part of the slab entered the trench approximately when the Japan Sea started to rift, (3) the tip of the horizontal slab is not the paleo-ridge between the Pacific and Izanagi Plates, and (4) some length of the Pacific slab has thus subducted before 30 Ma. This older subducted Pacific slab can now be seen in the lower mantle. Various tomographic models indeed show a steeply dipping high velocity anomaly that can be interpreted as a deep extension of the slab in the lower mantle, beneath the flat slab resting on the transition zone (Figures 6 and 7) but the resolution of these models does not allow concluding whether it is still attached or not.

This conclusion is in line with the reconstructions and models proposed by Seton et al. (2015) who showed that the kinematic reorganization in the Pacific is due to the subduction of the Pacific-Izanagi ridge and subsequent beginning of subduction of the Pacific Plate between 53 and 47 Ma. Figure 6c illustrates three time slices of the interactions between the Izanagi-Pacific Ridge at 70, 50, and 30 Ma based on Seton et al.'s (2012) kinematic model. As the ridge entered the subduction zone at about 50 Ma, the steeply dipping slab seen today in the lower mantle would thus be the remnant of the Izanagi Plate that subducted below Asia in the Mesozoic and Early Cenozoic, plus a part of the Pacific Plate. After ridge subduction, the Pacific Plate started to subduct some 50 Ma ago and slab pull progressively increased. After some 20 Ma, the slab pull component was large enough to trigger trench retreat and back-arc basins started to form. Because the Pacific-Izanagi ridge was roughly parallel to the margin, it subducted entirely along at the same time and back-arc basins thus started to form roughly simultaneously, around 30–35 Ma.

A similar geometry with a concave-up slab, compatible with trench retreat during the formation of back-arc basins, is observed all along the southern margin of Indonesia north of Sunda Trench (Figure 7; Hall & Spakman, 2015; Replumaz et al., 2004). This geometry suggests slab retreat accommodating the opening of basins in the overriding plate such as the South China Sea and other extensional basins (Rangin et al., 1990). This shows that the spreading of the eastern and southeastern margin of Asia since collision has affected the whole upper mantle, lithospheric mantle, and asthenosphere together, from the latitude of the Okhotsk and Japan Seas to Indonesia, and that the thin lithosphere below East and Southeast Asia rests above slabs ponding on the transition zone.

The transition from concave-up slabs to the overhanging and partly flat-lying Himalayan slab can be observed at the latitude of the Andaman Sea. The overhang increases northward toward the east Himalayan syntaxis (Figure 7). This dip reversal of the slab corresponds to the point separating the domains where the deep slab is observed, either north or south of the trench, right above the position where the high velocity anomaly in the lower mantle crosses the Sunda Trench (Figure 1; Hall & Spakman, 2015). East of this position the slab has retreated, and west of it, the slab has advanced, while India was indenting Asia. The Indian slab has thus followed the mantle flow that carries India northward and the mantle north of the slab has moved the same way, more than 3,000 km since 50 Ma. As the slab has moved with the flow, mantle flow lines should be continuous across the slab (Figure 7). To the southeast of the Andaman region, the slab has retreated toward the south and southwest, and the slab is continuous at depth; Indian mantle flow lines should then not be continuous across the trench.

## 7.3. Tectonics of Asia

We now discuss the large-scale deformation of Asia. As summarized above, the asymmetry of deformation in Asia is usually explained by the assumption that the western Pacific margin, because of subduction, is stress-

free and that the Asian continental lithosphere is free to move eastward, while it is blocked on the western side (Tapponnier et al., 1982). Although at the latitude of Japan the present-day subduction zone has been compressional for the last 10 Ma (Ingle, 1992; Jolivet et al., 1994; Tamaki et al., 1992), the whole eastern margin has been under extension from 30 to 10 Ma. It is still under extension today at the latitude of the Okinawa Basin in the upper plate of the Ruykyu subduction zone (Kimura, 1985; Sibuet et al., 1987).

Figures 8 and 9 (see also Figure S8) show the main active structures, today and at 25 Ma, together with the stress field and the pattern of seismic anisotropy at 50 and 200 km. The trend of the maximum horizontal stress ( $\sigma_{H-max}$ ) is shown smoothed from the World Stress Map (Heidbach et al., 2016; Zoback, 1992).  $\sigma_{H-min}$  is deduced from the orientation of extensional structures. On the 25 Ma reconstruction (Figure 9b), the directions of  $\sigma_{H-max}$  and  $\sigma_{H-min}$  are inspired from the present-day situation with local modifications to account for the orientation of the main extensional and strike-slip structures (Fournier et al., 1995). At 25 Ma, the India-Asia collision had been active for at least 25 Ma. The Tibetan Plateau and the Himalaya were under construction and the main large-scale strike-slip faults were already formed, so it appears plausible to assume that the stress field at large scales was similar to the present-day one.

These maps show that  $\sigma_{H-max}$  is well correlated with the mantle flow at 200 km (asthenosphere) and that  $\sigma_{H-min}$  better correlates with the direction of flow at 50 km. They also show that, at 25 Ma, the asthenospheric flow and  $\sigma_{H-max}$ , if resembling the present-day one, were oblique on the West Pacific subduction zone and compatible with the dextral shear recorded during the opening of the Japan Sea (Fournier et al., 2004).

## 7.3.1. Recent Compression at the Latitude of Japan Since 10 Ma

The reason for the change from extension to compression at the latitude of Japan is still unknown, but it could correspond to the progressive formation and displacement of the PHSP toward its present position and the stabilization of the Central Japan triple junction (Hall, 2002; Jolivet et al., 1989, 1994; Lallemand, 2016; Rangin et al., 1990). The insertion of the young PHSP between the Pacific and Eurasia Plates might have coupled the Pacific, the Philippine Sea, and Eurasia Plates (Jolivet et al., 1994). Faccenna et al. (2017) have recently proposed a complementary explanation. According to this study, the regular northward migration of PHSP progressively created a situation with two parallel subductions zones, the Ryukyu and Pacific Trenches, changing the kinematics of both trenches, from retreating to advancing, thus leading to a compressional tectonic regime at the latitude of Japan.

In the next subsection, we focus on the extension period that led to the opening of the Japan Sea, the South China Sea, and the Shikoku-Parece-Vela Basin on the PHSP, in the same Oligocene and Miocene time frame. **7.3.2. Back-Arc Basins Opening in the Oligocene and Miocene** 

During the Oligocene and part of the Miocene (Figure 9), the entire eastern and southeastern margin of Asia was extending behind retreating subduction zones with a slab progressively ponding on top of the transition zone. Lithospheric thinning associated with magmatism extends far west of the subduction zone, all the way to Mongolia (Hunt et al., 2012; Windley & Allen, 1993), and its timing is not always precisely constrained. Cenozoic extension was accommodated by a lithosphere that had already been thinned during the Mesozoic (Charles et al., 2013; Yin, 2010). Therefore, when India-Asia collision started, the eastern part of Asia was supported by a thin and warm lithosphere, and thinning continued until the Middle Miocene at least.

The cause for extension is classically attributed to slab retreat, assuming that it is able to transmit extensional forces at long distance inside Asia, as far as the Baikal Rift for instance, with a flow of asthenosphere toward the subduction zone, only due to subduction dynamics (Zhao, 2017). The flat slab over a long distance below the extended part of the Asian lithosphere is compatible with this mechanism.

The direction of opening of the Japan Sea and South China Sea was not perpendicular to the margin (Figures 2b and 9b), rather NW-SE in both cases, and it was associated with large-scale strike-slip faults, left-lateral faults radiating from the collision zone and dextral faults parallel to the margin (Jolivet et al., 1994; Kimura & Tamaki, 1986; Lallemand & Jolivet, 1985; Tapponnier et al., 1982; Yin, 2010). The geometry and kinematics of the Japan Sea back-arc opening was coeval with the development of a large-scale dextral shear zone, running from the northern Sea of Okhotsk to Southwest Japan along more than 2,000 km, which accommodated some 450 km of dextral displacement (Jolivet et al., 1994). The opening of the South China Sea, in the same direction as the Japan Sea, involved one large-scale sinistral shear zone, the Ailao Shan-Red River Shear Zone, running from the Tibetan Plateau to the region of Hanoi, and a dextral component





**Figure 8.** The main active structures in Asia and the western Pacific together with the principal horizontal stress trajectories (orange lines), smoothed from the World Stress Map (Heidbach et al., 2016; Zoback, 1992) and SL2013sv seismic anisotropy (red lines; Schaeffer & Lebedev, 2013) at 50 km (a) and 200 km (b).

along the western margin (Huchon et al., 1994; Marquis et al., 1997; Rangin et al., 1995; Tapponnier et al., 1982, 1986, 1990). Seismic profiles across the South China Sea rift in the non-oceanized domain show that the crust deformed ductilely during rifting (Clerc et al., 2018; Savva et al., 2013), suggesting a high heat flow and a thin and hot lithosphere.





**Figure 9.** Two time-shots (a: present-day, b: 25 Ma, similar to Figure 2b) of active structures with  $\sigma_{Hmax}$  and  $\sigma_{H-min}$ , on top of SL2013sv seismic anisotropy (50 km for the present-day stage, 200 km for the 25 Ma stage). As the main tectonic features of Asia were already present 25 Ma ago, stress trajectories on the reconstruction are taken from the present-day situation, offset southward to accommodate for the more southerly position of India at this period.  $\sigma_{H-min}$  is set perpendicular to  $\sigma_{Hmax}$  and compatible with the strike of extensional structures.

The 25 Ma reconstruction (Figure 9b) shows that the extension and related strike-slip faults at the latitude of Japan were oblique to the direction of the Pacific trench and that the asthenospheric flow (if it has not changed significantly, as suggested by the orientation of  $\sigma_{H-max}$ ) was also oblique on the margin, with a

configuration that is compatible with the observed dextral shear (Yin, 2010). However, the Sakhalin-Hokkaido dextral shear zone extends far north of the trench along the western margin of the Okhotsk Sea (Jolivet et al., 1999; Worrall et al., 1996), showing that the dextral shear is not only due to the obliquity of subduction. These dextral shear zones can also be far-field effects of the India-Asia collision as originally proposed by Kimura and Tamaki (1986) and confirmed by analogue modeling (Fournier et al., 2004; Jolivet et al., 1990, 1994; Kimura & Tamaki, 1986). A question then remains concerning the main driving forces behind a dynamic connection between back-arc extension and collision.

#### 7.4. Mantle Flow and Asian Tectonics

We have shown so far that the geometry of slabs below the Himalayas and the West Pacific back-arc basins and the direction of flow suggested by seismic anisotropy at 200 km and long-term kinematic trajectories are compatible with a single asthenospheric flow (Figure 7), with an advancing trench in the Himalaya and a retreating trench below the Japan Sea. This led to an oversteepened slab below the thick crust of the Himalayas and Tibet and concave-up slabs below the thinned lithosphere of eastern Asia.

Detailed studies of the stratification of seismic anisotropy beneath Tibet (Agius & Lebedev, 2017; Huang et al., 2017) show that the flow direction in the asthenosphere and the lithosphere can be quite different. The flow direction in the asthenosphere shown by these studies is similar to the flow at 200 km otherwise shown here based on a global model and the flow in the crust and lithosphere similar to the flow at 100 km. This is confirmed by the plot of the SL2013sv seismic anisotropy 25 km below LAB (Figure S5), similar to that at 200 km, except below Southeast Asia where the lithosphere is thin over a wide region. It is thus likely that the 100 and 200 km anisotropy inferences represent the strain or flow in the lithosphere and the asthenosphere, respectively.

Figure 7 shows a synthesis of mantle flow based on the observations summarized above. Black arrows represent the flow in the asthenosphere at 200 km, which is part of the whole mantle conveyor belt, and white arrows show the flow or strain at 100 km at lithospheric depths. Note that the divergence of the deep flow (200 km) toward the NE should induce a stretching perpendicular to flow lines, compatible with the direction of anisotropy in the lithosphere (100 km) and the direction of  $\sigma_{H-min}$ . In the vicinity of some slabs (SE Asia, western Mediterranean), the main flow seems mostly controlled by trench retreat. The flow pattern shown by white arrows indeed shows similar sharp bends in East Asia and the Middle East and eastern Mediterranean region (see inset on Figure 7a), with an opposite sense of rotation, clockwise versus counterclockwise, when approaching subduction zones. This toroidal flow can be interpreted as the result of the transition along strike of the convergence zone from collision to oceanic subduction and a large-scale slab tear (Capitanio, 2014; Funiciello et al., 2006; Magni et al., 2012, 2014; Menant, Sternai, et al., 2016; Sternai et al., 2014, 2016). Forward migration of the slab within the collision zone gives place to a retrograde motion of the oceanic slab and a return flow below the back-arc domain (Faccenna, et al., 2006). This back-arc domain is the Aegean Sea in the eastern Mediterranean and Southeast Asia east of the Himalayan collision zone. The anisotropy pattern at 50 or 100 km can be interpreted as the direction of extension (or flow) in the lithosphere, perpendicular to the main shortening direction. In Tibet, where gravitational potential energy is large, this results in the coexistence of reverse faulting at the periphery of the plateau, and strike-slip and normal faulting in the plateau itself. Farther away from the collision zone, deformation is dominated by extension and large-scale strike-slip shear zones.

#### 7.5. A Tectonic Scenario

As the Indian slab has moved northward since the onset of collision, the asthenosphere in which it sinks has moved at least by the same amount. The flow carrying India has thus reached north beyond the collision zone. We propose the following scenario: The asthenospheric flow carrying India northward has invaded the whole of the East Asian mantle, and spread out until the Pacific margin, where it has accommodated the Pacific slab retreat (Figures 7 and 10). We suggest here a long-distance coupling between collision and back-arc extension permitted by asthenospheric flow underneath Asia. This asthenospheric flow has thus been active below the whole collision zone. The forces controlling the distribution of deformation in this region (Tibetan Plateau, the eastern Himalayan syntaxis, Burma) then arise from several sources: (1) continent-continent collision transmits horizontal forces across the crust and lithosphere, and (2) gravitational potential energy stored in the thick crust tends to spread the thickened crust laterally (Molnar &



**Figure 10.** Cross-sections of the mantle showing the postulated flow pattern. Three stages of a possible evolution of the lithosphere and mantle along a section form South Africa to the western Pacific. Upwelling above LLSVP and downwelling in the lower mantle are partly disconnected from the horizontal flow in the upper mantle (gray line shows the 660 km discontinuity). The northward flow pushes India inside Asia and the Tethyan slab leaves detached fragments in the lower mantle behind, while the flow extends all the way to the western Pacific subduction zones, where it follows the Pacific slab retreat below Japan. The first two stages correspond to episodes of slab break-off in the Himalayas at 45 and 25 Ma and the last stage (10 Ma) corresponds to the transition from extension to compression along the eastern margin of the Japan Sea. The simplest possible scenario has been used for the Himalayas with only one slab progressively overridden and detaching and one collision event. More complex evolution could be envisaged. Lyon-Caen, 1988). Forces related to indentation induce the formation of strike-slip faults at long distance from the Himalaya (Molnar & Tapponnier, 1975; Tapponnier & Molnar, 1976). (3) Mantle flow underneath Asia is an additional driver that can explain the general asymmetry of deformation and its propagation eastward and southeastward.

During its northward motion, the Indian-Tethyan slab has left behind detached pieces that are now seen in tomographic models south of the collision zone, in the lower mantle (Figure 6a). The timing of this progressive dissection of the subducting slab is not precisely known, but geodynamic reconstructions suggest multiple shortening events and multiple break-off events between 45 and 25 Ma, ultimately leading to the complete loss of the oceanic slab and possibly two episodes of slab break-off at 45 and 25 Ma (Bouilhol et al., 2013; Mahéo et al., 2002, 2009; Replumaz et al., 2014, 2016). After slab break-off, the still attached part becomes free to move with the asthenospheric flow carrying India in the upper mantle. The asthenospheric flow would then continue toward the Pacific margin where the slab has started its retreat some 30–35 Ma ago, producing the shape we observe today (Figure 10). The part of the Indian slab lying horizontally below the eastern part of Tibet (Agius & Lebedev, 2013) might have been flattened by this northward flow (Magni et al., 2017). Figure 10 is based upon the simplest possible scenario for the Himalaya (one single collision event and one single slab, progressively overridden and detached) but a more complex evolution is possible.

This scenario explains in a simple way the geometry of the two slabs, the Indian slab that has overpassed the detached blobs it left in the lower mantle, and the Pacific slab that is resting with a shallow angle on the transition zone. The geometry of slabs is similar from the latitude of Japan to SE Asia, suggesting that the whole eastern Asian lithosphere has spread eastward and southeastward toward the retreating subduction zones (Figures 7 and 10).

In our scenario, deformation in the lithosphere is controlled by (1) a deep asthenospheric flow that results from the large-scale conveyor belt between a plume south of India and the west Pacific subductions and by (2) a shallower flow resulting from continent-continent collision and slab retreat east of the collision zone. In this context, the Tibetan Plateau formed the underthrusting of India and by body forces stored in the lithosphere resulting from crustal thickening and thinning of lithospheric mantle. Tibetan crustal material is extruded toward the east where it overrides the continental crust of Southeast China along the Long Men Shan thrust and toward the southeast toward the retreating subduction zone. Around the Eastern Himalayan Syntaxis and toward western Indochina, the uppermost mantle flows to the east and southeast, driven by the rollback of the Sunda slab, providing active support to the topography (Sternai et al., 2016). In this region, the deep NE-ward Indian asthenospheric flow is blocked by the Indian slab subducting below Indonesia. The flow passes through only north of the Andaman region where the slab is detached and overhanging.

This tectonic scenario is compatible with more recent models of collision dynamics in this region. Replumaz et al. (2016) have suggested, based on analogue modeling, that continental subduction does not necessarily require a strong slab pull. When the underthrusting plate is pushed from behind, the continent subducts even with a weak slab pull and the experiments reproduce the overturned shape of the Himalayan slab. In nature, the push force can be found in the entrainment of India by the underlying flow of the mantle. The interactions between subduction dynamics and upper plate deformation are also modulated by the intrinsic behavior of the subducting slab that has detached several times. Numerical experiments (Replumaz et al., 2014) have shown that slab break-off results in an efficient coupling of the two lithospheres through the plate contact and the propagation of compressive stresses far into the upper plate.

Assuming that during the formation of the Japan Sea the Japan Arc was transported by a combination of a deep eastward asthenospheric flow and a shallower southeastward lithospheric flow, and taking the amount of displacement (~450 km) along the Sakhalin-Hokkaido Shear Zone during this period (from 30 to 10 Ma; Jolivet et al., 1994) as a constraint, a simple velocity triangle can be constructed (Figure 4b). The direction of motion of Japan with respect to Eurasia is set parallel to the long axis of the Japan Sea, that is, parallel to the main strike-slip shear zone (NNE-SSW), while the two other directions are parallel (1) to the supposed mantle flow at 200 km and (2) the displacement of Japan with respect to the mantle. This leads to a velocity of the mantle flow at 200 km of about 2 cm/year eastward. When adding the rotation component of the Japan Arc to account for its bent shape (see Jolivet et al., 1994) its velocities with respect to Eurasia and to the mantle increase, but it does not affect significantly the velocity of the eastward mantle flow. With this

estimate, we assume that the eastern border of Eurasia is coupled to the mantle flowing underneath and it is thus a minimum value. Compared to the velocity of the mantle carrying India (~5 cm/year), only the half of it is preserved at the Pacific margin.

An apparent delay of  $\sim$ 20 Ma is observed between the India-Asia collision ( $\sim$ 50 Ma) and the inception of back-arc extension (~35–30 Ma), leading to the formation of the Japan Sea and South China Sea. But extension has been active along the eastern margin of Asia before and during the whole duration of the India-Asia collision, as soon as the Late Cretaceous (Maruyama et al., 1997; Yin, 2010). As shown in the synthesis of Pubellier and Morley (2014), extension has taken place in the overriding plate of the Sunda Trench in the Paleogene, forming distributed rift basins that preceded the formation of the South China Sea oceanic basin. In the same time period, extension was already active in the PHSP during its northward journey, forming the West Philippine Basin (Lallemand, 2016) in the Eocene before the Shikoku-Parece Vela Basin in the Oligo-Miocene and the Mariana basin since the Late Miocene. Whether this extension was a consequence of ridge subduction, associated with some limited slab retreat, and/or already included a component of eastward asthenospheric flow coming from the Tethyan convergence zone is unclear. The 20 Ma between collision and back-arc spreading initiation appear a necessary time lag until the Pacific slab was long and dense enough to induce a sufficient slab pull and allow slab retreat. This suggests that the interactions between collision and subduction through mantle flow below Asia works both ways: The mantle flow carrying India northward follows eastward slab retreat only when the latter is permitted by the negative buoyancy of the slab.

#### 7.6. Compatibility of our Scenario With Geochemical Anomalies

These geodynamic interactions between the Himalayan orogeny and the western Pacific are further supported by the evolution of the geochemical characteristics of magmatism in the circum-Pacific back-arc basins. In particular, one can consider the DUPAL-type anomaly that corresponds to anomalous Pb and Sr isotopic compositions of oceanic basalts first seen in the Indian Ocean (Dupré & Allègre, 1983). It was then shown that these anomalous compositions are organized with two maxima, closely linked with the regions now identified as LLSPVs the lower mantle (Castillo, 1988).

Our scenario is in line with work by Tamaki (1995) and Flower et al. (1998, 2001). Mostly based on geochemical data, these authors proposed that back-arc basins' formation in the western Pacific could result from an eastward *extrusion*, or eastward *wind* of the Asian mantle, in response to the continuous injection of mantle from the south below the India-Asia collision zone, thus explaining the DUPAL-like anomaly observed in magmatic rocks all the way to the West Pacific volcanic arcs. Tamaki (1995) and Flower et al. (1998, 2001) also suggested that this process could concern the back-arc basins carried by the PHSP and that the mantle wind could have made its eastward way through slab windows near Taiwan and the Philippines. The Shikoku-Parece Vela Basin (Figures 1a and 2a) is formed with the same timing as the South China Sea and the Japan Sea (Chamot-Rooke et al., 1987; Hall, 2002; Hall, Fuller, et al., 1995; Jolivet et al., 1989; Lallemand, 2016), and the Mariana Trough is still opening nowadays (Karig et al., 1978; Taylor, 1992). These basins, which are not under the influence of the deformation of the Asian lithosphere, developed parallel to the margin and do not show the obliquity shown by the Japan Sea or the South China Sea and the association with largescale strike-slip faults (Briais et al., 1993; Chamot-Rooke et al., 1987; Hall, 2002; Hall, Fuller, et al., 1995; Lallemand, 2016; Taylor & Hayes, 1983).

Indeed, it has long been recognized that the isotopic composition of the crust forming East Asian back-arc basins requires the involvement of an Indian-type mantle (Mukasa et al., 1987; Tatsumoto & Nakamura, 1991), a specific geochemical signature found in the Indian Ocean and Tethyan magmatic products (Bouilhol et al., 2010; Castillo, 1988; Mahoney et al., 1998). Whereas the Cretaceous basement of the western Philippine Sea Plate shows a Pacific and an Indian-type mantle signature (Hickey-Vargas et al., 2008), the compositions of Cenozoic and modern magmas clearly indicate an Indian-type mantle (Flower et al., 2001; Straub et al., 2009). Our model provides a possible explanation for the existence of a DUPAL-like anomaly in magmatic rocks of East Asia (Basu et al., 1991) with a high-tide limit along the Mariana and Bonin arcs (Flower et al., 1998, 2001). Liu et al. (2004) suggested that the distribution of low-velocity anomalies below Asia is consistent with such an extrusion of mantle and they showed with numerical models that the continuous injection of mantle underneath the India-Asia collision zone over the past 50 Ma could result in such an extrusion of mantle toward the east. Miyazaki et al. (2015) suggested instead that the boundary between the Indian and



Pacific mantles has been stationary in the western Pacific since the Cretaceous. The long-term subduction of the Tethys Ocean below the southern margin of Asia before collision may thus have allowed such an invasion on Indian mantle before collision started. In the interpretation of Flower et al. (1998, 2001), eastward mantle flow forced the slab to retreat. In pure kinematic terms, we propose a very similar tectonic scenario, but in dynamic terms we cannot separate forces due to slab retreat from those due to mantle extrusion.

## 8. Conclusion

Based on a comparison of kinematic trajectories spanning the last 50 Ma and the pattern of seismic anisotropy, we propose a scenario for the Cenozoic deformation of Asia. In this scenario, the deformation of Asia since the collision has been driven by (1) the asthenospheric flow diverging from the large low velocity anomaly present under South and West Africa and the Southwest Indian Ocean that reaches the West Pacific back-arc regions, (2) compressional stresses transmitted through the continent-continent collision zone in the lithosphere, and (3) slab retreat east and southeast of the collision zone.

After Tethys closure, the Indian slab has migrated northward by ~3,000 km carried by the same asthenospheric flow that reaches far beyond the collision zone, all the way to the Pacific margin of Asia. This flow also accommodates the extension recorded in eastern Asia, the opening of back-arc basins, and the retreat of the Pacific slab under Japan, and overrides the westward flow of Pacific mantle. A more superficial toroidal flow due to the lateral transition from continental collision in the Himalayas and oceanic subduction in the Sunda Trench induces a southward flow associated with retreat of Sunda Trench. The combination of these two flows leads to an oblique opening of back-arc basins, the Japan Sea and South China Sea, controlled by large-scale dextral and sinistral strike-slip shear zones. The main shortening direction in the deforming continent, between the collision zone and the Pacific subduction zones, could in this case be a proxy for the direction of flow in the asthenosphere underneath, which may become a useful tool for assessing mantle flow underneath continents in the distant past. This scenario of the India-Asia collision puts the emphasis on the role of the asthenosphere flowing underneath continents and may offer alternative ways of understanding tectonic processes.

#### Acknowledgments

The maps shown in this paper were produced with the Generic Mapping Tool (GMT: Wessel et al., 2013: Wessel & Smith, 1991). Thanks to Trond Torsvik who kindly provided his kinematic model and to Wim Spakman for the tomographic images. Special thanks are due to Romain Jolivet who kindly helped his father with GMT! This paper is a contribution of the ERC Advanced Research Grant RHEOLITH (Grant agreement No. 290864), of Institut Universitaire de France and Labex VOLTAIRE. TWB was partially support by NSF-EAR 1460479. P.S. was supported by the Swiss NSF (Ambizione grant PZ00P2\_168113/1). M.T. acknowledges the University of Trieste, Utrecht University, and the Netherlands Research Centre for Integrated Solid Earth Science (ISES) (ISES-2016-UU-19) for supporting this study. All data used in this paper are published.

### References

- Agius, M. R., & Lebedev, S. (2013). Tibetan and Indian lithospheres in the upper mantle beneath Tibet: Evidence from broadband surfacewave dispersion. *Geochemistry, Geophysics, Geosystems, 14*, 4260–4281. https://doi.org/10.1002/ggge.20274
- Agius, M. R., & Lebedev, S. (2017). Complex, multilayered azimuthal anisotropy beneath Tibet: Evidence for co-existing channel flow and pure-shear crustal thickening. *Geophysical Journal International*, 210(3), 1823–1844. https://doi.org/10.1093/gji/ggx1266
- Alvarez, W. (2010). Protracted continental collisions argue for continental plates driven by basal traction. *Earth and Planetary Science Letters*, 296(3–4), 434–442. https://doi.org/10.1016/j.epsl.2010.1005.1030
- Amaru, M. (2007). Global travel time tomography with 3-D reference models (PhD thesis, p. 174). Utrecht: Utrecht University.
- Artemieva, I. M., & Mooney, W. D. (2001). Thermal thickness and evolution of Precambrian lithosphere: A global study. Journal of Geophysical Research, 106B, 16,387–16,414.
- Barckhausen, U., Engels, M., Franke, D., Ladage, S., & Pubellier, M. (2014). Evolution of the South China Sea: Revised ages for breakup and seafloor spreading. *Marine and Petroleum Geology*, 58, 599–611. https://doi.org/10.1016/j.marpetgeo.2014.1002.1022
- Basu, A. R., Junwen, W., Wankang, H., Guanghong, X., & Tatsumoto, M. (1991). Major element, REE, and Pb, Nd and Sr isotopic geochemistry of Cenozoic volcanic rocks of eastern China: Implications for their origin from suboceanic-type mantle reservoirs. *Earth and Planetary Science Letters*, 393, 29–42.
- Becker, T. W., & Boschi, L. (2002). A comparison of tomographic and geodynamic mantle models. *Geochemistry, Geophysics, Geosystems, 3*(1), 1003. https://doi.org/10.1029/2001GC000168
- Becker, T. W., & Faccenna, C. (2011). Mantle conveyor beneath the Tethyan collisional belt. *Earth and Planetary Science Letters*, 310(3–4), 453–461. https://doi.org/10.1016/j.epsl.2011.1008.1021
- Becker, T. W., Lebedev, S., & Long, M. D. (2012). On the relationship between azimuthal anisotropy from shear wave splitting and surface wave tomography. *Journal of Geophysical Research*, *117*, B01306. https://doi.org/10.01029/02011JB008705
- Becker, T. W., & O'Connell, R. J. (2001). Predicting plate velocities with mantle circulation models. *Geochemistry, Geophysics, Geosystems*, 2(12), 1060. https://doi.org/10.1029/2001GC000171
- Becker, T. W., Schulte-Pelkum, V., Blackman, D. K., Kellog, J. B., & O'Connell, R. J. (2006). Mantle flow under the western United States from shear wave splitting. *Earth and Planetary Science Letters*, 247(3–4), 235–251. https://doi.org/10.1016/j.epsl.2006.05.010
- Bird, P. (1998). Testing hypotheses on plate-driving mechanisms with global lithosphere models including topography, thermal structure and faults. *Journal of Geophysical Research*, 103(B5), 10,115–10,129. https://doi.org/10.1029/98JB00198
- Bouilhol, P., Jagoutz, O., Hanchar, J. M., & Dudas, F. O. (2013). Dating the India–Eurasia collision through arc magmatic records. *Earth and Planetary Science Letters*, 366, 163–175. https://doi.org/10.1016/j.epsl.2013.1001.1023
- Bouilhol, P., Schaltegger, U., Chiaradia, M., Ovtcharova, M., Stracke, A., Burg, J. P., & Dawood, H. (2010). Timing of juvenile arc crust formation and evolution in the Sapat Complex (Kohistan–Pakistan). *Chemical Geology*, 280, 243–256.
- Boyden, J. A., Müller, R. D., Gurnis, M., Torsvik, T. H., Clark, J. A., Turner, M., et al. (2011). Next-generation plate-tectonic reconstructions using GPlates. In G. R. Keller & C. Baru (Eds.), *Geoinformatics: Cyberinfrastructure for the solid Earth sciences* (pp. 95–114). Cambridge, UK: Cambridge University Press. https://doi.org/10.1017/CB09780511976308.008

- Briais, A., Patriat, P., & Tapponnier, P. (1993). Updated interpretation of magnetic anomalies and seafloor spreading stages in the South China Sea: Implications for the tertiary tectonics of SE Asia. *Journal of Geophysical Research*, *98*, 6229–6328.
- Burke, K. (2011). Plate tectonics, the Wilson cycle, and mantle plumes: Geodynamics from the top. Annual Review of Earth and Planetary Sciences, 39(1), 1–29. https://doi.org/10.1146/annurev-earth-040809-152521
- Cammarano, F., Goes, S., Vacher, P., & Giardini, D. (2003). Inferring upper-mantle temperatures from seismic velocities. *Physics of the Earth and Planetary Interiors*, 138(3-4), 197–222. https://doi.org/10.1016/S0031-9201(03)00156-0
- Capitanio, F. A. (2014). The dynamics of extrusion tectonics: Insights from numerical modeling. *Tectonics*, 33(12), 2361–2381. https://doi.org/10.1002/2014TC003688
- Capitanio, F. A., Replumaz, A., & Riel, N. (2015). Reconciling subduction dynamics during Tethys closure with large-scale Asian tectonics: Insights from numerical modeling. *Geochemistry, Geophysics, Geosystems, 16*(3), 962–982. https://doi.org/10.1002/2014GC005660
- Castillo, P. (1988). The Dupal anomaly as a trace of the upwelling lower mantle. *Nature*, 336(6200), 667–670. https://doi.org/10.1038/ 336667a0
- Chamot-Rooke, N., Renard, V., & Pichon, X. L. (1987). Magnetic anomalies in the Shikoku Basin: A new interpretation. *Earth and Planetary Science Letters*, 83(1–4), 214–228. https://doi.org/10.1016/0012-821X(87)90067-7
- Chapman, M. C., & Solomon, S. C. (1976). North American-Eurasian plate boundary in Northeast Asia. Journal of Geophysical Research, 81(5), 921–930. https://doi.org/10.1029/JB081i005p00921
- Charles, N., Augier, R., Gumiaux, C., Monie, P., Chen, Y., Faure, M., & Zhu, R. (2013). Timing, duration and role of magmatism in wide rift systems: Insights from the Jiaodong Peninsula (China, East Asia). Gondwana Research, 24(1), 412–428. https://doi.org/10.1016/j. gr.2012.10.011
- Chen, Z., Schellart, W. P., Strak, V., & Duarte, J. C. (2016). Does subduction-induced mantle flow drive backarc extension? *Earth and Planetary* Science Letters, 441, 200–210. https://doi.org/10.1016/j.epsl.2016.1002.1027
- Clerc, C., Ringenbach, J. C., Jolivet, L., & Ballard, J. F. (2018). Rifted margins: Ductile deformation, boudinage, continentward-dipping normal faults and the role of the weak lower crust. *Gondwana Research*, *53*, 20–40. https://doi.org/10.1016/j.gr.2017.1004.1030
- Cobbold, P. R., & Davy, P. (1988). Indentation tectonics in nature and experiments. 2. Central Asia. Bulletin of the Geological Institutions of Upsalla, 14, 143–162.
- Coleman, M., & Hodges, K. (1995). Evidence for Tibetan Plateau uplift before 14 Myr ago from a new minimum age for east-west extension. *Nature*, 374(6517), 49–52. https://doi.org/10.1038/374049a0
- Conrad, C., & Lithgow-Bertelloni, C. (2006). Influence of continental roots and asthenosphere on plate-mantle coupling. *Geophysical Research Letters*, 33, L05312. https://doi.org/10.01029/02005GL025621
- Conrad, C. P., & Lithgow-Bertelloni, C. (2002). How mantle slabs drive plate tectonics. Science, 298(5591), 207–209. https://doi.org/10.1126/ science.1074161
- Copley, A., Avouac, J. P., & Royer, J. Y. (2010). India-Asia collision and the Cenozoic slowdown of the Indian plate: Implications for the forces driving plate motions. *Journal of Geophysical Research*, *115*, B03410. https://doi.org/10.01029/02009JB006634
- Dal Zilio, L., Faccenda, M., & Capitanio, F. (2017). The role of deep subduction in supercontinent breakup. *Tectonophysics*. https://doi.org/ 10.1016/j.tecto.2017.03.006
- Davy, P., & Cobbold, P. R. (1988). Indentation tectonics in nature and experiments. 1. Experiments scaled for gravity. Bulletin of the Geological Institutions of Upsalla, 14, 129–141.
- Debayle, E., Dubuffet, F., & Durand, S. (2016). An automatically updated S-wave model of the upper mantle and the depth extent of azimuthal anisotropy. *Geophysical Research Letters*, 43(2), 674–682. https://doi.org/10.1002/2015GL067329
- Dewey, J. F., Cande, S., & Pitman, W. C. I. (1989). Tectonic evolution of the India-Eurasia collision zone. *Eclogae Geologicae Helvetiae*, 82, 717–734.
- Doubrovine, P., Steinberger, B., & Torsvik, T. H. (2012). Absolute plate motions in a reference frame defined by moving hotspots in the Pacific, Atlantic and Indian oceans. Journal of Geophysical Research, 117, B09101. https://doi.org/10.01029/02011JB009072
- Dupré, B., & Allègre, C. J. (1983). Pb-Sr isotope variation in Indian Ocean basalts and mixing phenomena. *Nature*, 303(5913), 142–146. https://doi.org/10.1038/303142a0
- Faccenna, C., & Becker, T. W. (2010). Shaping mobile belts by small-scale convection. Nature, 465(7298), 602–605. https://doi.org/10.1038/ nature09064
- Faccenna, C., Becker, T. W., Auer, L., Billi, A., Boschi, L., Brun, J. P., et al. (2014). Mantle dynamics in the Mediterranean. *Reviews of Geophysics*, 52(3), 283–332. https://doi.org/10.1002/2013RG000444
- Faccenna, C., Becker, T. W., Conrad, C. P., & Husson, L. (2013). Mountain building and mantle dynamics. *Tectonics*, 32(1), 80–93. https://doi.org/10.1029/2012TC003176
- Faccenna, C., Becker, T. W., Jolivet, L., & Keskin, M. (2013). Mantle convection in the Middle East: Reconciling Afar upwelling, Arabia indentation and Aegean trench rollback. *Earth and Planetary Science Letters*, 375, 254–269. https://doi.org/10.1016/j. epsl.2013.1005.1043
- Faccenna, C., Bellier, O., Martinod, J., Piromallo, C., & Regard, V. (2006). Slab detachment beneath eastern Anatolia: A possible cause for the formation of the north Anatolian fault. *Earth and Planetary Science Letters*, 242(1–2), 85–97. https://doi.org/10.1016/j.epsl.2005.11.046

Faccenna, C., Holt, A. F., Becker, T. W., Lallemand, S., & Royden, L. H. (2017). Dynamics of the Ryukyu/Izu-Bonin-Marianas double subduction system. *Tectonophysics*. https://doi.org/10.1016/j.tecto.2017.1008.1011

- Faccenna, C., Piromallo, C., Crespo-Blanc, A., Jolivet, L., & Rossetti, F. (2004). Lateral slab deformation and the origin of the western Mediterranean arcs. *Tectonics*, 23, TC1012. https://doi.org/10.1029/2002TC001488
- Flesch, L. M., Holt, W. E., Silver, P. G., Stephenson, M., Wang, C. Y., & Chan, W. W. (2005). Constraining the extent of crust–mantle coupling in Central Asia using GPS, geologic, and shear wave splitting data. *Earth and Planetary Science Letters*, 238(1–2), 248–268. https://doi.org/ 10.1016/j.epsl.2005.06.023
- Flower, M., Tamaki, K., & Hoang, N. (1998). Mantle extrusion: A model for dispersed volcanism and DUPAL-like asthenosphere in East Asia and the western Pacific. In M. F. J. Flower, S.-L. Chung, C.-H. Lo, & T.-Y. Lee (Eds.), *Mantle dynamics and plate interactions in East Asia* (pp. 67–88). Washington, DC: American Geophysical Union. https://doi.org/10.1029/GD027p0067
- Flower, M. J. F., Russo, R. M., Tamaki, K., & Hoang, N. (2001). Mantle contamination and the Izu-Bonin-Mariana IBM 'high-tide mark': Evidence for mantle extrusion caused by Tethyan closure. *Tectonophysics*, 333(1–2), 9–34. https://doi.org/10.1016/S0040-1951(00)00264-X
- Fournier, M., Jolivet, L., Davy, P., & Thomas, J. C. (2004). Back arc extension and collision: An experimental approach of the tectonics of Asia. Geophysical Journal International, 157(2), 871–889. https://doi.org/10.1111/j.1365-246X.2004.02223.x
- Fournier, M., Jolivet, L., & Fabbri, O. (1995). Neogene stress field in SW Japan and mechanism of deformation during the Japan Sea opening. Journal of Geophysical Research, 12, 24,295–24,314.

- Fukao, Y., & Obayashi, M. (2013). Subducted slabs stagnant above, penetrating through, and trapped below the 660 km discontinuity. *Journal of Geophysical Research: Solid Earth*, *118*, 5920–5938. https://doi.org/10.1002/2013JB010466
- Funiciello, F., Moroni, M., Piromallo, C., Faccenna, C., Cenedese, A., & Bui, H. A. (2006). Mapping mantle flow during retreating subduction: Laboratory models analyzed by feature tracking. *Journal of Geophysical Research*, 111, B03402. https://doi.org/10.01029/02005JB003792

Ghosh, A., Becker, T. W., & Humphreys, E. D. (2013). Dynamics of the North American continent. *Geophysical Journal International*, 194(2), 651–669. https://doi.org/10.1093/gji/ggt1151

Griffin, W. L., O'Reilly, S. Y., Abe, N., Aulback, S., Davies, R. M., Pearson, N. J., et al. (2003). The origin and evolution of Archean lithospheric mantle. *Precambrian Research*, 127, 14–91.

Griggs, D. (1939). A theory of mountain building. American Journal of Science, 237(9), 611-650. https://doi.org/10.2475/ajs.237.9.611

Hall, R. (2002). Cenozoic geological and plate tectonic evolution of SE Asia and the SW Pacific: Computer-based reconstructions, model and animations. *Journal of Asian Earth Sciences*, 20(4), 353–431. https://doi.org/10.1016/S1367-9120(01)00069-4

Hall, R., Ali, J. R., & Anderson, C. D. (1995). Cenozoic motion of the Philippine Sea Plate: Palaeomagnetic evidence from eastern Indonesia. *Tectonics*, 14(5), 1117–1132. https://doi.org/10.1029/95TC01694

Hall, R., Ali, J. R., Anderson, C. D., & Baker, S. J. (1995). Origin and motion history of the Philippine Sea Plate. *Tectonophysics*, 251(1-4), 229–250. https://doi.org/10.1016/0040-1951(95)00038-0

Hall, R., Fuller, M., Ali, J. R., & Anderson, C. (1995). The Philippine Sea Plate, magnetism and reconstructions. In B. Taylor & J. Natland (Eds.), Active margins and marginal basins of the Western Pacific, Geophysical Monograph Series (Vol. 88, pp. 371–404). Washington, DC: American Geophysical Union.

Hall, R., & Spakman, W. (2015). Mantle structure and tectonic history of SE Asia. Tectonophysics, 658, 14–45. https://doi.org/10.1016/j. tecto.2015.1007.1003

Heidbach, O., Rajabi, M., Reiter, K., Ziegler, M., & WSM Team (2016). World stress map database release 2016. https://doi.org/10.5880/ WSM.2016.5001, GFZ Data Services.

Hickey-Vargas, R., Bizimis, M., & Deschamps, A. (2008). Onset of the Indian Ocean isotopic signature in the Philippine Sea Plate: Hf and Pb isotope evidence from Early Cretaceous terranes. *Earth and Planetary Science Letters*, 268(3–4), 255–267. https://doi.org/10.1016/j. epsl.2008.01.003

Holmes, A. (1931). Radioactivity and Earth movements. Transactions of the Geological Society of Glasgow, 18(3), 559–606. https://doi.org/ 10.1144/transglas.18.3.559

Honda, S. (2016). Slab stagnation and detachment under Northeast China. *Tectonophysics*, 671, 127–138. https://doi.org/10.1016/j. tecto.2016.1001.1025

Huang, Z., Tilmann, F., Xu, M., Wang, L., Ding, Z., Mi, N., et al. (2017). Insight into NE Tibetan Plateau expansion from crustal and upper mantle anisotropy revealed by shear-wave splitting. *Earth and Planetary Science Letters*, 478, 66–75. https://doi.org/10.1016/j.epsl.2017.08.030
Huchon, P., Le Pichon, X., & Rangin, C. (1994). Indochina Peninsula and the collision of India and Eurasia. *Geology*, 22(1), 27–30. https://doi.

org/10.1130/0091-7613(1994)022<0027:IPATCO>2.3.CO;2 Hunt, A. C., Parkinson, I. J., Harris, N. B. W., Barry, T. L., Rogers, N. W., & Yondon, M. (2012). Cenozoic volcanism on the Hangai Dome, Central Mongolia: Geochemical evidence for changing melt sources and implications for mechanisms of melting. *Journal of Petrology*, *53*(9), 1913–1942. https://doi.org/10.1093/petrology/egs038

Husson, L., Conrad, C. P., & Faccenna, C. (2012). Plate motions, Andean orogeny, and volcanism above the South Atlantic convection cell. *Earhth Planetary Science Letters*, 317–318, 126–135. https://doi.org/10.1016/j.epsl.2011.11.040

Ingle, C. J. (1992). Subsidence of the Japan Sea: Stratigraphic evidence from ODP sites and onshore sections. In K. Tamaki, K. Suyehiro, J. Allan, M. McWilliams, et al. (Eds.), Proc. ODP, Sci. Results (pp. 1197–1218). College Station, TX: Ocean Drilling Program.

Jolivet, L. (1986). America-Eurasia plate boundary in eastern Asia and the opening of marginal basins. *Earth and Planetary Science Letters*, 81, 282–288.

Jolivet, L., Davy, P., & Cobbold, P. (1990). Right-lateral shear along the Northwest Pacific margin and the India-Eurasia collision. *Tectonics*, 9(6), 1409–1419. https://doi.org/10.1029/TC009i006p01409

Jolivet, L., Faccenna, C., Agard, P., Frizon de Lamotte, D., Menant, A., Sternai, P., & Guillocheau, F. (2016). Neo-Tethys geodynamics and mantle convection: From extension to compression in Africa and a conceptual model for obduction. *Canadian Journal of Earth Sciences*, 53(11), 1190–1204. https://doi.org/10.1139/cjes-2015-0118

Jolivet, L., Faccenna, C., d'Agostino, N., Fournier, M., & Worrall, D. (1999). The kinematics of marginal basins, examples from the Tyrrhenian, Aegean and Japan Seas. In C. Mac Niocaill & P. D. Ryan (Eds.), *Continental tectonics* (pp. 21–53). London: Geological Society.

Jolivet, L., Faccenna, C., Goffé, B., Burov, E., & Agard, P. (2003). Subduction tectonics and exhumation of high-pressure metamorphic rocks in the Mediterranean orogens. *American Journal of Science*, 303(5), 353–409. https://doi.org/10.2475/ajs.2303.2475.2353

Jolivet, L., Faccenna, C., & Piromallo, C. (2009). From mantle to crust: Stretching the Mediterranean. *Earth and Planetary Science Letters*, 285(1–2), 198–209. https://doi.org/10.1016/j.epsl.2009.1006.1017

Jolivet, L., Huchon, P., Brun, J. P., Chamot-Rooke, N., Le Pichon, X., & Thomas, J. C. (1991). Arc deformation and marginal basin opening, Japan Sea as a case study. *Journal of Geophysical Research*, *96*(B3), 4367–4384. https://doi.org/10.1029/90JB02455

Jolivet, L., Huchon, P., & Rangin, C. (1989). Tectonic setting of western Pacific marginal basins. *Tectonophysics*, *160*(1–4), 23–47. https://doi. org/10.1016/0040-1951(89)90382-X

Jolivet, L., Tamaki, K., & Fournier, M. (1994). Japan Sea, opening history and mechanism, a synthesis. Journal of Geophysical Research, 99(B11), 22237–22259. https://doi.org/10.1029/93JB03463

Karato, S. (1993). Importance of anelasticity in the interpretation of seismic tomography. *Geophysical Research Letters*, 20(15), 1623–1626. https://doi.org/10.1029/93GL01767

Karig, D. E., Anderson, R. N., & Bibee, L. D. (1978). Characteristics of back arc spreading in the Mariana Trough. *Journal of Geophysical Research*, 83(B3), 1213–1226. https://doi.org/10.1029/JB083iB03p01213

Kimura, G., & Tamaki, K. (1986). Collision, rotation and back arc spreading: The case of the Okhotsk and Japan seas. *Tectonics*, 5(3), 389–401. https://doi.org/10.1029/TC005i003p00389

Kimura, M. (1985). Back arc rifting in the Okinawa Trough. Marine and Petroleum Geology, 2(3), 222–240. https://doi.org/10.1016/0264-8172(85)90012-1 Koptev, A., Burov, E., Calais, E., Leroy, S., Gerya, T., Guillou-Frottier, L., & Cloetingh, S. (2016). Contrasted continental rifting via plume-craton

interaction: Applications to Central East African Rift. *Geoscience Frontiers*, 7(2), 221–236. https://doi.org/10.1016/j.gsf.2015.1011.1002 Koptev, A., Calais, E., Burov, E., Leroy, S., & Gerya, T. (2015). Dual continental rift systems generated by plume-lithosphere interaction. *Nature Geoscience*, 8(5), 388–392. https://doi.org/10.1038/NGEO2401

Goes, S., Agrusta, R., van Hunen, J., & Garel, F. (2017). Subduction-transition zone inter- action: A review. *Geosphere*, 13(3), 1–21. https://doi. org/10.1130/GES01476.01471

Lallemand, S. (2016). Philippine Sea Plate inception, evolution, and consumption with special emphasis on the early stages of Izu-Bonin-Mariana subduction. *Progress in Earth and Planetary Science*, 3(1), 15. https://doi.org/10.1186/s40645-40016-40085-40646 Lallemand, S., & Jolivet, L. (1985). Japan Sea: A pull apart basin. *Earth and Planetary Science Letters*, 76, 375–389.

**Tectonics** 

Lavé, J., Avouac, J. P., Lacassin, R., Tapponnier, P., & Montagner, J. P. (1996). Seismic anisotropy beneath Tibet: Evidence for eastward extrusion of the Tibetan lithosphere ? *Earth and Planetary Science Letters*, 140(1-4), 83–96. https://doi.org/10.1016/0012-821X(96)00045-3

Little, T. A., Savage, M. K., & Tikoff, B. (2002). Relationship between crustal finite strain and seismic anisotropy in the mantle, Pacific–Australia plate boundary zone, South Island, New Zealand. *Geophysical Journal International*, *151*(1), 106–116. https://doi.org/10.1046/j.1365-246X.2002.01730.x

Liu, M., Cui, X., & Liu, F. (2004). Cenozoic rifting and volcanism in eastern China: A mantle dynamic link to the Indo–Asian collision? *Tectonophysics*, 393(1–4), 29–42. https://doi.org/10.1016/j.tecto.2004.07.029

Long, M. D., & Becker, T. W. (2010). Mantle dynamics and seismic anisotropy. Earth and Planetary Science Letters, 297(3-4), 341-354. https://doi.org/10.1016/j.epsl.2010.06.036

Magni, V., Allen, M. B., van Hunen, J., & Bouilhol, P. (2017). Continental underplating after slab break-off. *Earth and Planetary Science Letters*, 474, 59–67. https://doi.org/10.1016/j.epsl.2017.1006.1017

Magni, V., Faccenna, C., van Hunen, J., & Funiciello, F. (2014). How collision triggers backarc extension: Insight into Mediterranean style of extension from 3-D numerical models. *Geology*, 42(6), 511–514. https://doi.org/10.1130/G35446.35441

Magni, V., van Hunen, J., Funiciello, F., & Faccenna, C. (2012). Numerical models of slab migration in continental collision zones. Solid Earth, 3(2), 293–306. https://doi.org/10.5194/se-5193-5293-2012

Mahéo, G., Blichert-Toft, J., Pin, C., Guillot, S., & Pêcher, A. (2009). Partial melting of mantle and crustal sources beneath South Karakorum, Pakistan: Implications for the Miocene geodynamic evolution of the India-Asia convergence zone. *Journal of Petrology*, *50*(3), 427–449. https://doi.org/10.1093/petrology/egp006

Mahéo, G., Guillot, S., Blichert-Toft, J., Rolland, Y., & Pêcher, A. (2002). A slab breakoff model for the Neogene thermal evolution of South Karakorum and South Tibet. *Earth and Planetary Science Letters*, 195(1–2), 45–58. https://doi.org/10.1016/S0012-821X(01)00578-7

Mahoney, J. J., Frei, R., Tejada, M. L. G., Mo, X. X., Leat, P. T., & Nägler, T. F. (1998). Tracing the Indian Ocean mantle domain through time: Isotopic results from old west Indian, east Tethyan, and South Pacific seafloor. *Journal of Petrology*, 39(7), 1285–1306. https://doi.org/ 10.1093/petroj/39.7.1285

Marquis, G., Roques, D., Huchon, P., Coulon, O., Chamot-Rooke, N., Rangin, C., & Le Pichon, X. (1997). Amount and timing of extension along the continental margin off Central Vietnam. Bulletin de la Societe Geologique de France, 168(6), 707–716.

Maruyama, S., Isozaki, Y., Kimura, G., & Terabayashi, M. (1997). Paleogeographic maps of the Japanese Islands: Plate tectonic synthesis from 750 Ma to the present. *The Island Arc*, 6(1), 121–142. https://doi.org/10.1111/j.1440-1738.1997.tb00043.x

McDonough, W. F., & Sun, S. S. (1995). The composition of the Earth. Chemical Geology, 120(3–4), 223–253. https://doi.org/10.1016/0009-2541(94)00140-4

McKenzie, D. P. (1969). Speculations on the consequences and causes of plate motions. *Geophysical Journal of the Royal Astronomical Society*, 18(1), 1–32. https://doi.org/10.1111/j.1365-246X.1969.tb00259.x

Menant, A., Jolivet, L., & Vrielynck, B. (2016). Kinematic reconstructions and magmatic evolution illuminating crustal and mantle dynamics of the eastern Mediterranean region since the late Cretaceous. *Tectonophysics*, 675, 103–140. https://doi.org/10.1016/j.tecto.2016.1003.1007

Menant, A., Sternai, P., Jolivet, L., Guillou-Frottier, L., & Gerya, T. (2016). 3D numerical modeling of mantle flow, crustal dynamics and magma genesis associated with slab roll-back and tearing: The eastern Mediterranean case. *Earth and Planetary Science Letters*, 442, 93–107. https://doi.org/10.1016/j.epsl.2016.1003.1002

Miyazaki, T., Kimura, J. I., Senda, R., Vaglarov, B. S., Chang, Q., Takahashi, T., et al. (2015). Missing western half of the Pacific Plate: Geochemical nature of the Izanagi-Pacific Ridge interaction with a stationary boundary between the Indian and Pacific mantles. *Geochemistry*, *Geophysics. Geosystems*. 16, 3309–3332. https://doi.org/10.1002/2015GC005911

Molnar, P., & Lyon-Caen, H. (1988). Some simple physical aspects of the support, structure, and evolution of mountain belts. *Geological Society of America Special Papers*, 218, 179–207. https://doi.org/10.1130/SPE218-p179

Molnar, P., & Lyon-Caen, H. (1989). Fault plane solutions of earthquakes and active tectonics of the Tibetan plateau and its margins. *Geophysical Journal International*, 99(1), 123–154. https://doi.org/10.1111/j.1365-246X.1989.tb02020.x

Molnar, P., & Tapponnier, P. (1975). Cenozoic tectonics of Asia: Effects of a continental collision. Science, 189(4201), 419–426. https://doi.org/ 10.1126/science.189.4201.419

Mukasa, S. B., McCabe, R., & Gill, J. B. (1987). Pb-isotopic compositions of volcanic rocks in the West and East Philippine island arcs: Presence of the Dupal isotopic anomaly. *Earth and Planetary Science Letters*, 84(2–3), 153–164. https://doi.org/10.1016/0012-1821X(1087)90082-90083

Obayashi, M., Yoshimitsu, J., Nolet, G., Fukao, Y., Shiobara, H., Sugioka, H., et al. (2013). Finite frequency whole mantle *P* wave tomography: Improvement of subducted slab images. *Geophysical Research Letters*, 40, 5652–5657. https://doi.org/10.1002/2013GL057401

Pownall, J. M., Lister, G. S., & Spakman, W. (2017). Reconstructing subducted oceanic lithosphere by "reverse-engineering" slab geometries: The northern Philippine Sea Plate. *Tectonics*, 36, 1814–1834. https://doi.org/10.1002/2017TC004686

Pubellier, M., & Morley, C. K. (2014). The basins of Sundaland (SE Asia): Evolution and boundary conditions. *Marine and Petroleum Geology*, 58, 555–578. https://doi.org/10.1016/j.marpetgeo.2013.1011.1019

Rangin, C., Huchon, P., Bellon, H., Le Pichon, X., Lepvier, C., Roques, D., et al. (1995). Cenozoic tectonics of central and South Vietnam. *Tectonophysics*, 251(1–4), 179–196. https://doi.org/10.1016/0040-1951(95)00006-2

Rangin, C., Jolivet, L., & Pubellier, M. (1990). A simple model for the tectonic evolution of Southeast Asia and Indonesia region for the past 43 my. *Bulletin de la Societe Geologique de France*, *6*, 889–906.

Rangin, C., Le Pichon, X., Mazzotti, S., Pubellier, M., Chamot-Rooke, N., Aurelio, M., et al. (1999). Plate convergence measured by GPS across the Sundaland/Philippine Sea Plate deformed boundary: The Philippines and eastern Indonesia. *Geophysical Journal International*, 139(2), 296–316. https://doi.org/10.1046/j.1365-246x.1999.00969.x

Rangin, C., Maurin, T., & Masson, F. (2013). Combined effects of Eurasia/Sunda oblique convergence and East-Tibetan crustal flow on the active tectonics of Burma. Journal of Asian Earth Sciences, 76, 185–194. https://doi.org/10.1016/j.jseaes.2013.1005.1018

Replumaz, A., Capitanio, F. A., Guillot, S., Negredo, A. M., & Villaseñor, A. (2014). The coupling of Indian subduction and Asian continental tectonics. *Gondwana Research*, 26(2), 608–626. https://doi.org/10.1016/j.gr.2014.1004.1003

Li, C., van der Hilst, R., Engdahl, E. R., & Burdick, S. (2008). A new global model for *P* wave speed variations in Earth's mantle. *Geochemistry, Geophysics, Geosystems, 9*, Q05018. https://doi.org/10.01029/02007GC001806

Liu, X., Zhao, D., Li, S., & Wei, W. (2017). Age of the subducting Pacific slab beneath East Asia and its geodynamic implications. Earth and Planetary Science Letters, 464, 166–174. https://doi.org/10.1016/j.epsl.2017.1002.1024



Replumaz, A., Funiciello, F., Reitano, R., Faccenna, C., & Balon, M. (2016). Asian collisional subduction: A key process driving formation of the Tibetan Plateau. *Geology*, 44(11), 943–946. https://doi.org/10.1130/G38276.1

Replumaz, A., Guillot, S., Villaseñor, A., & Negredo, A. M. (2013). Amount of Asian lithospheric mantle subducted during the India/Asia collision. Gondwana Research, 24(3–4), 936–945. https://doi.org/10.1016/j.gr.2012.1007.1019

Replumaz, A., Karason, H., van der Hilst, R., Besse, J., & Tapponnier, P. (2004). 4-D evolution of SE Asia's mantle from geological reconstructions and seismic tomography. *Earth and Planetary Science Letters*, 221(1–4), 103–115. https://doi.org/10.1016/S0012-1821X(1004)00070–00076

Ritsema, J., Deuss, A., van Heijst, H. J., & Woodhouse, J. H. (2011). S40RTS: A degree-40 shear-velocity model for the mantle from new Rayleigh wave dispersion, teleseismic traveltime and normal-mode splitting function measurements. *Geophysical Journal International*, 184(3), 1223–1236. https://doi.org/10.1111/j.1365-246X.2010.04884.x

Rohrmann, A., Kapp, P., Carrapa, B., Reiners, P. W., Guynn, J., Ding, L., & Heizler, M. (2012). Thermochronologic evidence for plateau formation in Central Tibet by 45 Ma. *Geology*, 40(2), 187–190. https://doi.org/10.1130/G32530.32531

Royden, L. H., Burchfiel, B. C., & van der Hilst, R. D. (2008). The geological evolution of the Tibetan Plateau. *Science*, 321(5892), 1054–1058. https://doi.org/10.1126/science.1155371

Runcorn, S. K. (1962). Towards a theory of continental drift. Nature, 193(4813), 311-314. https://doi.org/10.1038/193311a0

Savage, M. K. (1999). Seismic anisotropy and mantle deformation: What have we learned from shear wave splitting? *Reviews of Geophysics*, 37(1), 65–106. https://doi.org/10.1029/98RG02075

Savva, D., Meresse, F., Pubellier, M., Chamot-Rooke, N., Lavier, L., Po, K. W., et al. (2013). Seismic evidence of hyper-stretched crust and mantle exhumation offshore Vietnam. *Tectonophysics*, 608, 72–83. https://doi.org/10.1016/j.tecto.2013.1007.1010

Schaeffer, A. J., & Lebedev, S. (2013). Global shear-speed structure of the upper mantle and transition zone. *Geophysical Journal International*, 194(1), 417–449. https://doi.org/10.1093/gji/ggt1095

Schaeffer, A. J., Lebedev, S., & Becker, T. W. (2016). Azimuthal seismic anisotropy in the Earth's upper mantle and the thickness of tectonic plates. *Geophysical Journal International*, 207(2), 901–933. https://doi.org/10.1093/gji/ggw1309

Schellart, W. P., Jessell, M. W., & Lister, G. S. (2003). Asymmetric deformation in the backarc region of the Kuril arc, northwest Pacific: New insights from analogue modeling. *Tectonics*, 22(5), 1047. https://doi.org/10.1029/2002TC001473

Seton, M., Flament, N., Whittaker, J., Müller, D., Gurnis, M., & Bower, D. J. (2015). Ridge subduction sparked reorganization of the Pacific platemantle system 60–50 million years ago. *Geophysical Research Letters*, 42, 1732–1740. https://doi.org/10.1002/2015GL063057

Seton, M., Müller, R. D., Zahirovic, S., Gaina, C., Torsvik, T., Shephard, G., et al. (2012). Global continental and ocean basin reconstructions since 200 Ma. *Earth-Science Reviews*, 113(3–4), 212–270. https://doi.org/10.1016/j.earscirev.2012.1003.1002

Sibuet, J. C., Letouzey, J., Barbier, F., Charvet, J., Foucher, J. P., Hilde, T. W. C., et al. (1987). Back-arc extension in the Okinawa Trough. Journal of Geophysical Research, 92(B13), 14,041–14,063. https://doi.org/10.1029/JB092iB13p14041

Simmons, N. A., Myers, S. C., Johannesson, G., & Matzel, E. (2012). LLNL-G3Dv3: Global P wave tomography model for improved regional and teleseismic travel time prediction. *Journal of Geophysical Research*, *117*, B10302. https://doi.org/10.11029/12012JB009525

Smith, S. G., Wegmann, K. W., Ancuta, L. D., Gosse, J. C., & Hopkins, C. E. (2016). Paleotopography and erosion rates in the central Hangay Dome, Mongolia: Landscape evolution since the mid-Miocene. *Journal of Asian Earth Sciences*, 125, 37–57. https://doi.org/10.1016/j. jseaes.2016.1005.1013

Spakman, W., & Wortel, R. (2004). A tomographic view on western Mediterranean geodynamics. In W. Cavazza, F. M. Roure, W. Spakman, G. M. Stampfli, & P. A. Ziegler (Eds.), *The TRANSMED atlas. The Mediterranean region from crust to mantle* (pp. 31–52). Berlin, Heidelberg: Springer. https://doi.org/10.1007/978-3-642-18919-7\_2

Steinberger, B., & Becker, T. W. (2016). A comparison of lithospheric thickness models. *Tectonophysics*. https://doi.org/10.1016/j. tecto.2016.1008.1001

Sternai, P., Avouac, J. P., Jolivet, L., Faccenna, C., Gerya, T., Becker, T. W., & Menant, A. (2016). On the influence of the asthenospheric flow on the tectonics and topography at a collision-subduction transition zones: Comparison with the eastern Tibetan margin. *Journal of Geodynamics*, *100*, 184–197. https://doi.org/10.1016/j.jog.2016.1002.1009

Sternai, P., Jolivet, L., Menant, A., & Gerya, T. (2014). Subduction and mantle flow driving surface deformation in the Aegean-Anatolian system. Earth and Planetary Science Letters, 405, 110–118. https://doi.org/10.1016/j.epsl.2014.1008.1023

Stoddard, P. R., & Abbott, D. (1996). Influence of the tectosphere upon plate motion. Journal of Geophysical Research, 101(B3), 5425–5433. https://doi.org/10.1029/95JB03540

Straub, S. M., Goldstein, S. L., Class, C., & Schmidt, A. (2009). Mid-ocean-ridge basalt of Indian type in the northwest Pacific Ocean basin. *Nature Geoscience*, 5(5), 304–289. https://doi.org/10.1038/NGEO1471

Tamaki, K. (1995). Upper mantle extrusion tectonics of southeast Asia and formation of western Pacific backarc basins, International Workshop: Cenozoic Evolution of the Indochina Peninsula (89 pp.). Hanoi/Do Son, Vietnam.

Tamaki, K., Suyehiro, K., Allan, J., Ingle, J. C., & Pisciotto, K. (1992). Tectonic synthesis and implications of Japan Sea ODP drilling. In Proc. ODP, Sci. Results (pp. 1333–1350). College Station, TX: Ocean Drilling Program.

Tapponnier, P., Lacassin, R., Leloup, P. H., Schärer, U., Dalai, Z., Haiwei, W., et al. (1990). The Ailao Shan/Red River metamorphic belt: Tertiary left-lateral shear between Indochina and South China. *Nature*, 343(6257), 431–437. https://doi.org/10.1038/343431a0

Tapponnier, P., & Molnar, P. (1976). Slip line field theory and large-scale continental tectonics. Nature, 264(5584), 319–324. https://doi.org/ 10.1038/264319a0

Tapponnier, P., & Molnar, P. (1977). Active faulting and tectonics in China. Journal of Geophysical Research, 82(20), 2905–2930. https://doi.org/ 10.1029/JB082i020p02905

Tapponnier, P., Peltzer, G., & Armijo, R. (1986). On the mechanics of the collision between India and Asia. In M. P. Coward & A. C. Ries (Eds.), *Collision tectonics. Geol. Soc. Spec. Pub., Geological Society, London* (Vol. 19, pp. 115–157).

Tapponnier, P., Peltzer, G., Dain, A. Y. L., Armijo, R., & Cobbold, P. (1982). Propagating extrusion tectonics in Asia: New insights from simple experiments with plasticine. *Geology*, *10*(12), 611–616. https://doi.org/10.1130/0091-7613(1982)10<611:PETIAN>2.0.CO;2

Tapponnier, P., Zhiqin, X., Roger, F., Meyer, B., Arnaud, N., Wittlinger, G., & Jingsui, Y. (2001). Oblique stepwise rise and growth of the Tibet Plateau. *Science*, 294(5547), 1671–1677. https://doi.org/10.1126/science.105978

Tatsumoto, M., & Nakamura, Y. (1991). DUPAL anomaly in the Sea of Japan: Pb, Nd, and Sr isotopic variations at the eastern Eurasian continental margin. *Geochimica et Cosmochimica Acta*, 55(12), 3697–3708. https://doi.org/10.1016/0016-7037(91)90068-G

Taylor, B. (1992). Rifting and the volcanic-tectonic evolution of the Izu-Bonin-Mariana arc. In B. Taylor & K. Fujioka (Eds.), Proc. ODP, Sci. Results (Vol. 126, pp. 627–651). College Station, TX: Ocean Drilling Program.

Taylor, B., & Hayes, D. E. (1983). Origin and history of the South China Sea basin. In D. E. Hayes (Ed.), The tectonic and geologic evolution of southeast Asian seas and islands, part 2 (pp. 23–56). Washington, DC: American Geophysical Union. https://doi.org/10.1029/GM027p0023

Tesauro, M., Kaban, M. K., & Cloetingh, S. A. P. L. (2013). Global model for the lithospheric strength and effective elastic thickness. *Tectonophysics*, 602, 78–86. https://doi.org/10.1016/j.tecto.2013.01.006

Tikoff, B., Russo, R., Teyssier, C., & Tommasi, A. (2004). Mantle-driven deformation of orogenic zone and clutch tectonics. In J. Grocott, K. J. W. McCaffrey, G. Taylor, & B. Tikoff (Eds.), *Vertical coupling and decoupling in the lithosphere* (pp. 41–64). London: Geological Society.

Torsvik, T. H., & Cocks, L. R. M. (2016). *Earth history and palaeogeography* (p. 536). Cambridge: Cambridge University Press. Torsvik, T. H., Van der Voo, R., Preeden, U., Mac Niocaill, C., Steinberger, B., Doubrovine, P. V., et al. (2012). Phanerozoic polar wander,

palaeogeography and dynamics. *Earth-Science Reviews*, 114(3–4), 325–368. https://doi.org/10.1016/j.earscirev.2012.1006.1007 van der Beek, P., Van Melle, J., Guillot, S., Pêcher, A., Reiners, P. W., Nicolescu, S., & Latif, M. (2009). Eocene Tibetan Plateau remnants preserved in the northwest Himalaya. *Nature Geoscience*, 5(7), 463–467. https://doi.org/10.1038/ngeo1503

van der Meer, D. G., Spakman, W., van Hinsbergen, D. J. J., Amaru, M. L., & Torsvik, T. H. (2010). Towards absolute plate motions constrained by lower-mantle slab remnants. *Nature Geoscience*, 6(2), 117–120. https://doi.org/10.1038/NGEO1708

Van der Voo, R., Spakman, W., & Bijwaard, H. (1999). Tethyan subducted slabs under India. Earth and Planetary Science Letters, 171(1), 7–20. https://doi.org/10.1016/S0012-821X(99)00131-4

van Hinsbergen, D. J. J., Kapp, P., Dupont-Nivet, G., Lippert, P. C., DeCelles, P. G., & Torsvik, T. H. (2011). Restoration of Cenozoic deformation in Asia and the size of Greater India. *Tectonics*, 30, TC5003. https://doi.org/10.1029/2011TC002908

Wang, C. Y., Flesch, L. M., Chang, L., & Zheng, T. (2013). Evidence of active mantle flow beneath South China. Geophysical Research Letters, 40(19), 5137–5141. https://doi.org/10.1002/grl.50987

Wang, C. Y., Flesch, L. M., Silver, P. G., Chang, L. J., & Chan, W. W. (2008). Evidence for mechanically coupled lithosphere in central Asia and resulting implications. *Geology*, *36*(5), 363–366. https://doi.org/10.1130/G24450A

Warners-Ruckstuhl, K. N., Meijer, P. T., Govers, R., & Wortel, M. J. R. (2010). A lithosphere-dynamics constraint on mantle flow: Analysis of the Eurasian plate. *Geophysical Research Letters*, 37, L18308. https://doi.org/10.11029/12010GL044431

Wessel, P., & Smith, W. H. F. (1991). Free software helps map and display data. Eos, Transactions American Geophysical Union, 72(441), 445–446.

Wessel, P., Smith, W. H. F., Scharroo, R., Luis, J. F., & Wobbe, F. (2013). Generic mapping tools: Improved version released. *Eos, Transactions American Geophysical Union*, 94(45), 409–410. https://doi.org/10.1002/2013EO450001

Williams, H., Turner, S., Kelley, S., & Harris, N. (2001). Age and composition of dikes in Southern Tibet: New constraints on the timing of eastwest extension and its relationship to postcollisional volcanism. *Geology*, 29(4), 339–342. https://doi.org/10.1130/0091-7613(2001)029<0339:AACODI>2.0.CO;2

Wilson, J. T. (1973). Mantle plume and plate motions. Tectonophysics, 19(2), 149-164. https://doi.org/10.1016/0040-1951(73)90037-1

Windley, B. F., & Allen, M. B. (1993). Mongolian Plateau: evidence for a Late Cenozoic mantle plume under central Asia. Geology, 21(4), 295–298. https://doi.org/10.1130/0091-7613(1993)021<0295:MPEFAL>2.3.CO;2

Worrall, D. M., Kruglyak, V., Kunst, F., & Kusnetsov, V. (1996). Tertiary tectonics of the Sea of Okhotsk, Russia: Far-field effects of the India-Asia collision. *Tectonics*, 15(4), 813–826. https://doi.org/10.1029/95TC03684

Yamato, P., Husson, L., Becker, T. W., & Pedoja, K. (2013). Passive margins getting squeezed in the mantle convection vice. *Tectonics*, 32(6), 1559–1570. https://doi.org/10.1002/2013TC003375

Yin, A. (2010). Cenozoic tectonic evolution of Asia: A preliminary synthesis. *Tectonophysics*, 488(1–4), 293–325. https://doi.org/10.1016/j. tecto.2009.1006.1002

Yu, C., Shi, X., Yang, X., Zhao, J., Chen, M., & Tang, Q. (2017). Deep thermal structure of Southeast Asia constrained by S-velocity data. Marine Geophysical Research, 38, 341–355. https://doi.org/10.1007/s11001-017-9311-x

Yu, Y., Gao, S. S., Liu, K. H., Yang, T., Xue, M., Le, K. P., & Gao, J. (2018). Characteristics of the mantle flow system beneath the Indochina Peninsula revealed by teleseismic shear wave splitting analysis. *Geochemistry, Geophysics, Geosystems*, 19(5), 1519–1532. https://doi.org/ 10.1002/2018GC007474

Zahirovic, S., Müller, R. D., Seton, M., & Flament, N. (2015). Tectonic speed limits from plate kinematic reconstructions. Earth and Planetary Science Letters, 418, 40–52. https://doi.org/10.1016/j.epsl.2015.1002.1037

Zhang, P. Z., Shen, Z., Wang, M., Gan, W., Bürgmann, R., Molnar, P., et al. (2004). Continuous deformation of the Tibetan Plateau from global positionning system data. *Geology*, 32(9), 809–812. https://doi.org/10.1130/G20554.20551

Zhao, D. (2017). Big mantle wedge, anisotropy, slabs and earthquakes beneath the Japan Sea. *Physics of the Earth and Planetary Interiors*, 270, 9–28. https://doi.org/10.1016/j.pepi.2017.1006.1009

Zhao, D., Isozaki, Y., & Maruyama, S. (2017). Seismic imaging of the Asian orogens and subduction zones. *Journal of Asian Earth Sciences*, 145, 349–367. https://doi.org/10.1016/j.jseaes.2017.1006.1016

Zoback, M. L. (1992). First- and second-order patterns of stress in the lithosphere: The world stress map project. *Journal of Geophysical Research*, 97(B8), 11,703–11,728. https://doi.org/10.1029/92JB00132